

Middlesex University Research Repository:

an open access repository of Middlesex University research

http://eprints.mdx.ac.uk

Zhang, Tiantian, 1990. The use of macrophytes for heavy metal pollution control in urban wetlands. Available from Middlesex University's Research Repository.

Copyright:

Middlesex University Research Repository makes the University's research available electronically.

Copyright and moral rights to this thesis/research project are retained by the author and/or other copyright owners. The work is supplied on the understanding that any use for commercial gain is strictly forbidden. A copy may be downloaded for personal, non-commercial, research or study without prior permission and without charge. Any use of the thesis/research project for private study or research must be properly acknowledged with reference to the work's full bibliographic details.

This thesis/research project may not be reproduced in any format or medium, or extensive quotations taken from it, or its content changed in any way, without first obtaining permission in writing from the copyright holder(s).

If you believe that any material held in the repository infringes copyright law, please contact the Repository Team at Middlesex University via the following email address: eprints@mdx.ac.uk

The item will be removed from the repository while any claim is being investigated.

THE USE OF MACROPHYTES FOR HEAVY METAL POLLUTION CONTROL IN URBAN WETLANDS

TIANTIAN ZHANG

A thesis submitted to the Council for National Academic Awards in partial fulfilment of the requirements for the Degree of Master of Philosophy

> Urban Pollution Research Centre Middlesex Polytechnic Bounds Green Road, London, N11 2NQ

> > October 1990

The Use of Macrophytes for Heavy Metal Pollution Control in Urban Wetlands

Tiantian Zhang

ABSTRACT

A literature survey has been undertaken to identify the role and function of aquatic macrophytes in the removal of contaminants from polluted surface waters. The survey focussed primarily on heavy metals, however nitrate, phosphate and hydrocarbon removal are also discussed and the application of aquatic macrophytes in wastewater treatment systems is also described.

The objectives of the research are: To determine heavy metal variations in water, sediment and plant tissues in both field and greenhouse conditions and to assess the management implications of macrophytes for heavy metal pollution control in urban wetlands.

The sampling sites and the methods selected for the determination of heavy metal levels in water column, sediment and plant tissues are also discussed.

Heavy metal (Cu, Pb, Zn and Cd) uptake rates and the maximum storage capabilities by sediment and plant have been investigated in controlled greenhouse-based phytoassay experiments. Typha latifolia heavy metal uptake ability is in the order of Cd>Cu>Zn>Pb.

Temporal variations of Cu, Pb, Zn and Cd in urban stormwater runoff, sediment and <u>Typha latifolia</u> tissues at the selected field sites have also been investigated. An increase in sediment and <u>Typha</u> tissue levels is observed from rural to urban sites. Field samples from all sites show a progressive decrease in metal levels from sediment to root to rhizome to leaf. All metals show temporal variability in Typha tissues.

<u>Typha latifolia</u> biomass and plant metal distributions have also been investigated. The rhizome of <u>Typha</u> can store 54 - 61 % and the leaves 32 - 40 % of the total metals found accumulated in the plant. An annual crop of <u>Typha</u> leaf biomass can remove up to Cu 525, Pb 415, Zn 680 and Cd 59 g/ha. Leaf harvesting could provide a useful method of heavy metal removal from recently constructed wetlands.

LIST OF CONTENTS

Abstract

List of Contents

	Page No.
<u>Chapter 1:</u> Introduction	1
1.1 The Research Context	1
1.2 The Metals Selected for Investigation	2
1.3 The Plants Selected for Investigation	3
1.4 Thesis Content and Organisation	5
<u>Chapter 2:</u> A Review of the Use of Aquatic	7
Macrophytes in Water Pollution	
Control	
2.1 Introduction	7
2.2 Heavy Metal Removal	8
2.2.1 Sources of Heavy Metals	8
2.2.2 Heavy Metals in the Aquatic Environment	9
2.2.3 Heavy Metal Speciation	11
2.2.4 The Toxicity of Heavy Metals	12
2.2.5 Heavy Metal Uptake by Macrophytes	15
2.2.6 The Effects of Heavy Metal Uptake	17
on Macrophytes	
2.2.7 The Mechanisms of Heavy Metals Uptake	20
2.2.8 Heavy Metal Removal	22
2.2.9 Heavy Metal Release	23
2.2.10 Biological Accumulation of Heavy Metals	24
in the Marine Environment	

2.3 Nutrient Pollution Removal	26
2.3.1 The Effects of Nutrient Uptake	27
2.3.2 Nutrient Storage and Release	28
2.4 Hydrocarbon Pollution Removal	29
2.4.1 Hydrocarbon Degradation	29
2.4.2 Oil Clean-up by Plants	30
2.5 The Application of Aquatic Macrophytes as	31
Wastewater Treatment Systems	
2.5.1 Use of Water Hyacinth in Wastewater	31
Treatment	
2.5.2 The Root Zone Biotechnology System	34
of Wastewater Treatment	
<u>Chapter 3:</u> Locations and Methods	39
3.1 Field Study	39
3.1.1 Sampling Locations	39
3.1.2 Sampling Methods	46
3.1.3 Laboratory Methods	46
3.1.4 Biomass Measurement	50
3.2 Greenhouse Experiments	50
3.2.1 Short Term Heavy Metal Dosing Experiments	50
3.2.2 Long Term Heavy Metal Dosing Experiments	51
<u>Chapter 4:</u> Dosing Experiments	55
4.1 Introduction	55
4.2 Discussion of the Results	56
4.2.1 Short Term Dosing Experiments	56
4.2.2 Long Term Dosing Experiments	59
4.2.2.1 Long Term Dosing Experiments with	59
<u>Typha latifolia</u>	

4.2.2.2 Long Term Dosing Experiments with	67
Juncus effusus	
4.2.2.3 Long Term Dosing Experiment with Peat	70
4.2.2.4 Comparison of Dosing Experiments with	72
Typha latifolia and Juncus effusus	
4.3 Conclusions	75
Chapter 5: Field Studies of Metal Uptake by	77
Typha latifolia	
5.1 Introduction	77
5.2 Discussion of the Results	78
5.2.1 Heavy Metal Levels in Wetland Water Samples	78
5.2.2 Heavy Metal Levels in Wetland Sediments	82
5.2.3 Heavy Metal Levels in <u>Typha</u> Root Tissues	86
5.2.4 Heavy Metal Levels in <u>Typha</u> Rhizome Tissues	90
5.2.5 Heavy Metal Levels in <u>Typha</u> Leaf Tissues	93
5.2.6 Comparison of Heavy Metal Levels in	97
Water, Sediment and Plant Tissues	
5.3 Conclusions	101
Chapter 6: The Comparison of Field Studies and	103
Greenhouse Dosing Experiments and	
their Management Implications	
6.1 Introduction	103
6.2 The Comparison of Field Studies and Greenhouse Experiments	103
6.2.1 The Comparison of Sediment Metal Levels	103
6.2.2 The Comparison of Typha Tissue Metal Levels	106
6.2.3 Metal Uptake Efficiency of Typha latifolia	109
6.3 Management Implications for Heavy Metal Removal	109

6.3.1 Biomass Measurement	109
6.3.2 Metal Distribution in Plant Tissues	111
6.3.3 Heavy Metal Removal	113
6.4 Conclusions	114
<u>Chapter 7:</u> Conclusions	116
7.1 Summary of Major Findings	116
7.2 Suggestions for Further Work	118
Acknowledgements	120
References	121

Appendix A

LIST OF FIGURES

<u>Chapter 2:</u>	A Review of the Use of Aquatic Macrophytes in Water Pollution Control	7
Figure 2.1	The Flows of Energy and Matter	10
Figure 2.2	Seasonal Variation in Concentrations of Zn, Cu and Pb in Leaves and Stems of	19
Figure 2.3	<u>Phragmites australis</u> Helophyte Bed Designs	35
Chapter 3:	Locations and Methods	39
Figure 3.1	Sampling Locations in North London	40
Figure 3.2	Site 1 The Dew Pond Situated in Trent Park	41
Figure 3.3	Sites 2 (Stonyfield Park) and 3 (Welsh Harp)	43
Chapter 4:	Dosing Experiments	55
Figure 4.1	Heavy Metal Levels in Peat and <u>Typha</u> Tissues	57
Figure 4.2	Metal Level Variations in Surface and Bottom Peat Samples During Typha latifolia	61
	8 Week Dosing Experiments	
Figure 4.3	Temporal Increases in Peat and Typha	63
TTEATE 4.7	Tissues Metal Concentrations During 8	
	Week Doging Fynariments	
Figure 4 4	Reck Dusting Experiments Recreasion Equations and Craphs for Matal	64
rigure 4.4	Negression Equations and Graphs for Metal	04
,	opeake by reat and <u>typna</u> itssues builing	

8	Week	Dosing	Experiments
---	------	--------	-------------

Figure 4.5	Temporal Increases in Peat and <u>Juncus</u>	68
	Tissue Metal Concentrations During 8	
	Week Dosing Experiments	
Figure 4.6	Regression Equations and Graphs of Metal	69

- Uptake by Peat and <u>Juncus</u> Tissues during 8 Week Dosing Experiments
- Figure 4.7 Metal Levels in Surface and Bottom Peat in 71 the Presence or Absence of <u>Typha</u> During 8 Week Dosing Experiments
- Figure 4.8 Metal Levels in <u>Typha latifolia</u> 73 and <u>Juncus effusus</u> During 8 Week Dosing Experiments

<u>Chapter 5:</u> Field Studies of Metal Uptake by 77 <u>Typha latifolia</u>

Figure	5.1	Average and Ranges of Heavy Metal	79
		Concentration in Wetland Water Samples	
Figure	5.2	Temporal Variations of Heavy Metal Levels	81
		in Wetland Water Samples	

- Figure 5.3 Average and Ranges of Heavy Metal 82 Concentration in Wetland Sediment
- Figure 5.4 Temporal Variations of Heavy Metal Levels 85 in Wetland Sediment Figure 5.5 Average and Ranges of Heavy Metal 86 Concentration in Typha Root Tissues
- Figure 5.6 Temporal Variations of Heavy Metal Levels 89 in <u>Typha</u> Root Tissues Figure 5.7 Average and Ranges of Heavy Metal 90
- Figure 5.7 Average and Kanges of Heavy Helai 90 Concentration in <u>Typha</u> Rhizome Tissues Figure 5.8 Temporal Variations of Heavy Metal Levels 92
- Figure 5.8 Temporal Variations of Heavy Metal Levels 92 in <u>Typha</u> Rhizome Tissues

Figure 5.9 Average and Ranges of Heavy Metal	94
Concentration in <u>Typha</u> Leaf Tissues	
Figure 5.10 Temporal Variations of Heavy Metal Levels	96
in <u>Typha</u> Leaf Tissues	
Figure 5.11 Heavy Metal Levels in Sediment and	99
<u>Typha</u> Tissues	
Chapter 6: The Comparison of Field Studies and	103

		Greenhouse Dosing Experiments and	
		their Management Implications	
	<i>с</i> л		100
Figure	6.1	The Comparison of Typha latitolia	108
		Tissues Metal Levels in Dosing	
		Experiments and Field Studies	
Figure	6.2	Metal Load Distributions in <u>Typha</u> Tissues	112

LIST OF TABLES

.

<u>Chapter 2:</u> A Review of the Use of Aquatic	7
Macrophytes in Water Pollution Control	
Table 2.1 Classification of metals Associated with Dusts, Sediments and Soils	12
Table 2.2 The Background and the Toxic Levels of Heavy Metals	13
Table 2.3 Maximum Heavy Metal Removal Values by Water Hyacinth	16
Table 2.4 Comparative Increases of Concentrations of Zn in <u>Typha latifolia</u>	17
Table 2.5 Concentration Factors in Benthic Algae Tissue and Dissolved Metals in Water	25
Table 2.6 Storage of Nitrogen and Phosphorus and Rate of Plant Uptake for Selected Aquatic Macrophytes	28
Table 2.7 Oxygen Transport through Selected Aquatic Macrophytes	30
Table 2.8 Effect of Growing Water Hyacinth in Partially Digested Liquor	33
Table 2.9 Processes for the Transformation of Vascular Aquatic Plants into Industrial Products	38
<u>Chapter 3:</u> Locations and Methods	39
Table 3.1 Parameters Used for Model 384 Polarographic Analyser	48
Table 3.2 Dosing Frequency and Water Volume Used	52

for Dosing Experiment with Typha	
Table 3.3 Dosing Frequency and Water Volume Used	53
for Dosing Experiment with <u>Juncus</u>	
Table 3.4 Dosing Frequency and Water Volume Used	54
for Dosing Experiment with Peat	
Chapter 4: Dosing Experiments	55
Table 4.1 Ratios of Peat, Root, Rhizome and Leaf	58
Metal Levels Relative to Control Peat	
Table 4.2 Metal Concentration Ratios of Surface to	60
Bottom Peat During Experiment with <u>Typha</u>	
Table 4.3 Metal Loads in <u>Typha</u> Tissues for	66
Dosing Experiment	
Table 4.4 Comparison of Metal Root and Leaf Ratios	74
between Controlled and Dosed Plant	
Chapter 5: Field Studies of Metal Uptake by	77
<u>Typha latifolia</u>	
Table 5.1 T Values of Water Samples	79
Using Students 't' Test	
Table 5.2 Average and Standard Deviation of	80
Metal Levels in Water Samples	
Table 5.3 T Values of Sediment Samples	83
Using Students 't' Test	
Table 5.4 Average and Standard Deviation of	84
Metal Levels in Sediment Samples	
Table 5.5 T Values of Typha Root Samples	87
Using Students 't' Test	
Table 5.6 Average and Standard Deviation of	88

Metal Levels in <u>Typha</u> Root Samples	
Table 5.7 T Values of Typha Rhizome Samples	91
Using Students 't' Test	
Table 5.8 Average and Standard Deviation of	93
Metal Levels in Typha Rhizome Samples	
Table 5.9 T Values of Typha Leaf Samples	94
Using Students 't' Test	
Table 5.10 Average and Standard Deviation of	95
Metal Levels in Typha Leaf Samples	
Table 5.11 Metal Level Ratios of Sediment	9 8
and Plant to Water	
Table 5.12 Sediment to Leaf Metal Average	100
Concentration Ratio	
Chapter 6: The Comparison of Field Studies and	103
Greenhouse Dosing Experiments and	
Table 6.1 The Toxicity of Metal Levels in Sediment	104
Table 6.2 Maximum Metal Loads in Sediment for	106
Dosing Experiments and Field Studies	
Table 6.3 The Toxicity of Metal Levels in Plant	1 07
Table 6.4 Dry Weight Tissue Biomass	110
Table 6.4 Dry Weight Tissue Biomass Table 6.5 Dry Weight Leaf Tissue Biomass	110 110

.

LIST OF PLATES

Page No.

Chapter	<u>3:</u> Locations and Methods	39
Plate 3.	1 Site 1 The Dew Pond Situated in Trent Park	42
Plate 3.	2 Site 2 The Ornamental Pond Situated	45
	in Stonyfield Park	
Plate 3.	,3 Site 3 The Northern Arm of the	45
	Welsh Harp Reservoir	
Plate 3.	4 Philips SP9 Atomic Absorption	49
	Spectrophotometer Used for Sediment	
	and Plant Analysis	
Plate 3.	.5 Model 384 Polarographic Analyser	49
	Used for Water Analysis	

Chapter 1: Introduction

1.1 The Research Context

The potential polluting effects of metals in surface runoff have been recognised for a long time (Carpenter, 1924; Jones, 1935). The adverse effects of such discharges on urban aquatic environments have also been documented (Van Hassel et al., 1979; May and Mckinney, 1981). Metal contamination of receiving waters has been identified as the cause of both fisheries and faunal depletion (Mance, 1987) and also can present hazards to public health (Kobayshi, 1971).

The most frequently adopted 'single-shot' solution to the runoff problem is to provide on-site storage urban facilities which potentially can offer the dual purpose function of both quantity and quality control (Hall et al, 1990). Whilst the large and highly variable volumes presented to such storage basins will generally render unfeasible any application of the sort of treatment processes used for sanitary or industrial effluents, the detention and biofiltration possibilities of combined provide considerable least-cost contact treatment for urban stormwater management and opportunities of experience root zone control. The pollution provide the unique (RZB) systems may biotechnology advantage of enabling primary filtration and secondary biological uptake for both solid and soluble pollutant phases through the judicious use of macrophytic planting programmes (Cooper et al., 1989). RZB systems are discussed further in Chapter 2 where their role as pollution scavengers is developed in further detail.

Even though COD, BOD and nutrient removal by RZB systems have been extensively studied, the role of macrophytes in heavy metal uptake and removal has not been fully investigated. The study of the efficiency and pathways whereby metal removal by macrophytes occurs in stormwater runoff might therefore potentially provide a useful method for heavy metal pollution control in urban wetlands. It is within this general pollution context that the current research project has been specifically devised and developed.

The research has been carried out with the following major objectives in mind:

a) To determine heavy metal variations in urban stormwater runoff, sediment and plant tissues across selected field sites and to establish relationships between these 'carrier' phases.

b) To determine heavy metal uptake efficiency and maximum storage capabilities of plants under controlled greenhouse-based phytoassay experimental conditions.

c) To assess the management implications of macrophytes for heavy metal pollution control in urban wetlands.

1.2 The Metals Selected for Investigation

The concentrations and loadings of copper (Cu), lead (Pb), zinc (Zn) and cadmium (Cd) in urban stormwater runoff, sediment (or peat in the case of greenhouse experiments), plant root, rhizome and leaf tissues were investigated during this research project. It was impossible to analyse a further range of heavy metals in the time required and

- 2 -

therefore this study has been focused on the abovementioned metals.

Cu, Pb, Zn and Cd comprise those metals which present the greatest overall hazard to the aquatic environment especially to fish and vertebrates (Dryssen et al., 1971, Mance, 1987). Lead is a widely used metal and a major vehicle derived pollutant which is a neurotoxin and has given rise to much public concern (Jaworski, 1979; USEPA, 1980). It is associated in particular with the urban environment (WHO, 1989). Cadmium is a highly toxic metal which can cause kidney disease in humans and this has stimulated considerable scientific attention (Kobayshi, 1971; Friberg et al., 1972). Copper and Zinc are also widely used metals in the urban industrial environment and are therefore commonly encountered pollutants in urbanised (Mance, 1987). Although catchments both metals are essential elements for plant, animal and human nutrition, they are toxic at certain threshold levels to many aquatic organisms.

1.3 The Plants Selected for Investigation

Typha latifolia, an aquatic macrophyte, was selected as the main species for the study project with Juncus effusus greenhouse experiments. Typha also being used for monitor metal uptake in selected to latifolia was account of its wide preference to Phragmites sp on distribution and abundance in urban wetlands and its ability to survive transplanting to laboratory conditions.

<u>Typha latifolia</u> is a native emergent species which grows in shallow water, especially on inorganic substrates or where there is silting and rapid decay of organic matter,

- 3 -

in lakes, ponds, canals and slow-flowing rivers. It is widely distributed in suitable habitats throughout Britain, though it occurs less frequently in the north. Typha species colonize new sites by seed dispersal, with seedling establishment being largely restricted to areas of exposed wet mud (Valk and Davis, 1976; Grace, 1984). Seeds are very small, disperse great distances by wind, and are produced in abundance with values of 222,000 seeds per inflorescence reported (Yeo, 1964). Typha latifolia is a good pre-emptive competitor when the plants are larger low densities (Grace, 1987), and has a great and at capacity for vegetative spread (McNaughton, 1975) and a great proportion of its biomass occurs below ground. One plant of Typha can spread by its rhizome over areas as great as 60 m^2 within 2 years of germination (Dykyjova and Kvet, 1978). After a habitat is fully colonized, seed germination may be inhibited (Grace, 1984; Grace and Harrison, 1986), and the stand is maintained by vegetative reproduction. In some marshes the Typha is not rooted in the mineral soil but floats on a fibrous mat. It is thought that buoyancy of the mat is mainly due to the Typha rhizome having a lower specific gravity than water and to gases that are trapped in the floating mat. The habitat conditions on this floating mat appear to change little over long periods and in time Typha may eventually form almost pure stands, expand, and coalesce to form a complete cover (Krusi and Wein, 1988). Typha latifolia a great ability to survive transplanting. plants have after transplanting showed that by means of Studies rhizome and shoot clumps, the growth of the plants is excellent (Kyte, 1984).

Juncus effusus is a British native soft rush. It is very abundant and locally dominant in wet pastures, bogs and

- 4 -

damp woods, especially on acid soils. This plant was selected for this research study because of the persistence of its foliage during the winter and its ability to survive transplanting.

1.4 Thesis Content and Organisation

The structure of the thesis and a fuller description of the research programme is outlined below. Following this introductory section, Chapter 2 provides a review of the which gives further literature and more detailed background information for the research project. The major sources and the toxicity of heavy metals found in urban wetlands along with the mechanisms of plant heavy metal uptake, removal and release are described. The use of macrophytes for nutrient and hydrocarbon removal and the application of aquatic macrophytes as wastewater treatment systems are also briefly discussed. Chapter 3 provides the details of sampling locations, sampling methods and analytical techniques which have been employed in the research.

Chapters 4 and 5 detail the two main aspects of the undertaken of the work in the course experimental describes the greenhouse dosing 4 Chapter research. experiments which investigate and analyse both peat and uptake efficiency and maximum storage plant metal Chapter 5 describes the investigation capabilities. levels undertaken to determine heavy metal in urban stormwater runoff, sediment and plant tissues under field conditions and the relationships established between them. Chapter 6 compares the results of greenhouse dosing experiments and field studies and develops the management implications of macrophytes for heavy metal pollution control. The discussions presented in Chapter 4, 5 and 6 also form part of two co-authored publications entitled "Metal Uptake and Associated Pollution Control by <u>Typha</u> <u>latifolia</u> in Urban Wetlands" (Zhang et al, 1990), and "The Use of Macrophytes for Water Pollution Control in China and UK" (Zhang et al, 1990). These articles are included as appendices to the thesis. The concluding Chapter 7 provides a general overview of the main conclusions and suggestions for further work.

<u>Chapter 2:</u> A Review of the Use of Aquatic Macrophytes in Water Pollution Control

2.1 Introduction

Aquatic macrophytes have been extensively studied over the last few decades in terms of their ability to take up and absorb pollutants, and therefore indirectly provide indication of water quality (Haslam, 1982; an Wang, 1986). It has been found that many species of aquatic macrophytes can grow rapidly and exhibit a tolerance of polluted water (Haslam and Wolseley, 1981). They can take up a variety of pollutants quickly and can be easily harvested, thereby providing both a temporal measure of pollution and a method of removing pollutants from the aquatic system. A number of studies have been carried out during the last 15 years defining the use of aquatic macrophytes in water pollution control (Mortimer, 1985; Reddy and DeBusk, 1987). In particular, recent research has been directed toward the use of aquatic macrophytes for wastewater treatment (Middlebrooks and Reed, 1981; Arthur, 1986; Cooper et al., 1989).

This review focuses on the ability of particular species of aquatic macrophytes to remove contaminants from polluted surface waters. The principle focus is on heavy metals, but other pollutants such as nitrate, phosphate and hydrocarbon removal are briefly discussed and the application of aquatic macrophytes as wastewater treatment systems is also described.

2.2. Heavy Metal Removal

2.2.1 Sources of Heavy Metals

The sources of heavy metals found within urban wetlands are both natural and anthropogenic. Natural sources include weathering of rocks, impervious surfaces and soils. The anthropomorphic influences include industrial activities, such as smelting, metal working; mining and metal extraction processes (Lagerwerff and Brower, 1975; Davies and Ginnever, 1979); domestic sources, such as paint, house dust, printed matter, paper, textiles and plastics (Sturges and Harrison, 1985) and vehicle exhausts which are particularly evident in the urban environment (Harrison, 1979).

The contributing sources of heavy metals to wetlands include suspended river loads; storm runoff; sewer atmospheric deposition discharges and (Ellis, 1976: Eaton, 1979; Revitt and Ellis, 1980). Street dusts often contain elevated concentrations of a range of toxic metals (Sartor and Boyd, 1972) which can be discharged from surface water sewers to urban wetland environments. In some instances these dusts represent a significant pollutant source, especially when stormwater runoff removes the street materials and their associated metal from the roadway causing an increased metal input to rivers (Harrison et al., 1981; Ellis and Harrop, 1984).

2.2.2 Heavy Metals in the Aquatic Environment

The flows of energy and matter in an aquatic ecosystem are shown in Fig. 2.1. In such aquatic systems, heavy metals are partitioned between different compartments of the ecosystem such as water, sediment and biota (Moore and Ramamoorthy, 1984; Prahalad and Seenayya, 1989). In water, which is the primary contaminated medium, a high degree of variation in metal concentration arises (Forstner and Wittman, 1979). Metals that do not remain soluble in water are sorbed and accumulated by bottom sediments acting as a sink (Hakanson, 1980). Sediments play a crucial role in water quality. Although they may remove pollutants from the water column, pollutants which have accumulated in the sediments may provide the surface water (long after the source of pollution has ceased) with contaminants (Salomons, 1985). The content of heavy metals in contaminated sediment is normally much higher than in water and plants (Ellis, 1986). The amount of heavy metals present in the sediments may thus be considered to be the amount potentially available to the plants.



Fig.2.1 The Flows of Energy and Matter in Aquatic Ecosystems

10 -

1

2.2.3 Heavy Metal Speciation

The environmental mobility and bioavailability of a metal is dependent upon the physico-chemical forms in which the metal is associated with sediment and water. The classifications of metals associated with dusts, sediments and soils are given in Table 2.1. A physicospeciation chemical scheme has been applied to the analysis of Cu, Pb, Zn and Cd in urban stormwater (Morrison et al., 1984). According to this study, Zn and Cd exhibit a preference for the dissolved phase whereas Pb predominates in the suspended solid phase. Copper tends to be distributed equally between both phases. The potentially toxic forms of the metals, (the bioavailable metals) in the dissolved phase (electrochemically available) and in the exchangeable particulate phase are said to account for about 63% of the total Zn, 77% of the total Cd, 66% of the total Pb and 32% of the total Cu.

In overall terms the results of Harrison et al.'s study street dusts chemical associations in and (1981) of the mobility soils suggest that and roadside bioavailability of the four metals probably declines in the following order: Cd>Pb Zn>Cu, whereas Ellis et al., (1986) suggests that the overall availability of street surface sediment associated metals to the dissolved phase of runoff is Cd>Zn, Pb>Cu.

Table 2.1: Classification of Metals Associated with Dusts, Sediment, and Soils (from Harrison et al., 1981)

classification	form of association*	extraction technique		
soluble	metal ppt; pore water	release to pure water or river water		
exchangeable	specifically adsorbed, ion exchangeable	exchange with excess cations		
carbonate phase	ppt or co-ppt	release by mild acid		
Fe-Mn oxide	specifically adsorbed,	reduction		
phase	co-ppt			
organic phase	complexed; adsorbed	oxidation		
residual	in mineral lattices	digestion with strong acid		

*ppt = precipitate.

2.2.4 The Toxicity of Heavy Metals

The concentrations of all elements in soil will be reflected to some extent in the dry matter of plants which grow in them (Baker and Walker, 1989). High levels of heavy metals in sediment and plant can be toxic. The background and toxic metal concentrations of soil and plants are given in Table 2.2. It has been found that soil pollution by heavy metals can restrict the growth of plants. A reduction of biomass production and nutritional quality is observed in crops grown in soil contaminated with moderate levels of heavy metals (Cottenie et al., 1976; Lepp, 1981). These elements generally inhibit physiological processes, e.g. photosynthesis (Carlson et al., 1975; Clijsters and Van Assche, 1985), phloem translocation and transpiration (Carlson et al., 1975); respiration is less sensitive (Lee et al., 1977b; Van Assche et al., 1979).

Zinc. Zn has long been known as an essential element in plant nutrition. It acts either as a metal component of enzymes (such as alcohol dehydrogenase, superoxide dismutase, carbonic anhydrase and RNA polymerase) or as a functional, structural or regulatory co-factor of a large number of enzymes (Price et al., 1972; Jyung, 1975; Marschner, 1986).

	Soils (mg/kg))	Plants (mg/kg)			
	Unpolluted	toxic level	Unpolluted	toxic level		
Pb	25	100	10	150		
Cd	1	5	-	5		
Cu	25	100	12	40		
Zn	65	300	60	400		

Table 2.2: The Background and the Toxic Levels of Heavy Metals (from Romero et al., 1989; Baker and Brooks, 1989)

High Zn levels in soil (or sediment) can be toxic to the plants. Zn toxicity results in a reduction in root and shoot growth(Van Assche et al., 1988, Allus et al., 1988) and leaf expansion which is followed by chlorosis. High levels of Zn in the nutrient medium depress the uptake of P and Fe (Adriano et al., 1971).

Copper. Cu is also an essential plant micronutrient being a component of the protein structure of a range of enzymes involved in electron transport and redox reactions in mitochondria, chloroplasts, cell walls and the cytoplasm of plant cells. Other Cu proteins play important roles in carbohydrate and nitrogen metabolism, lignification of the cell wall (Boardman, and 1975; Marschner, 1986).

For most plant species high amounts of Cu in the nutrient medium are toxic to growth. The effect appears to relate in part to the ability of Cu to displace other metal ions and particularly Fe from physiologically important centres. Chlorosis is thus a commonly observed symptom of Cu toxicity (Daniels et al., 1972). The inhibition of root growth is one of the most rapid responses to toxic Cu levels (Wainwright and Woolhouse, 1975).

Lead. Lead is neither an essential nor beneficial element in plant nutrition. Pb is a major chemical pollutant of the environment and is highly toxic to man. The reason for Pb toxicity in plants is not clear. However, the organic form trialkyl Pb is a powerful mutagenic agent and is known to derange the spindle fibre mechanism of cell division in plant cells (Bryce-Smith, 1975). **Cadmium.** Cd is toxic to plants. The basic cause of the toxicity probably lies in the much higher affinity of Cd for thiol groupings in enzymes and other proteins. The presence of Cd therefore disturbs enzyme activity (Mengel and Kirkby, 1978).

2.2.5 Heavy Metal Uptake by Macrophytes

Most macrophyte species can take up heavy metals and some of them can accumulate heavy metals to high levels (Adams et al., 1973; Tokunaga et al., 1976; Chigbo et al., 1982; Sridhar, 1986; Lan et al., 1990). Chigbo et al. (1982) found that the exposure of the water hyacinth (Eichhornia crassipes) to Cd levels of 10 mg/l for 2 days resulted in stem concentration of 501 mg/kg and а а leaf concentration of 574 mg/kg. Muramoto and Oki (1983)reported that after exposure to a 8.0 mg/l Pb solution days, the Pb levels in water hyacinth roots for 16 reached 25790 mg/kg and 1810 mg/kg in the tops of the plant. The maximum heavy metal removal values by per individual plant are shown in Table 2.3.

Cu bioaccumulation in water lilies (Nuphar lutea) has been shown to reach levels of up to 110 mg/kg in petioles according to a field study carried out by Aulio (1980). Blake et al. (1987) reported that after one month of contact with sediment (initial concentration of Zn in the medium was 10 mg/kg), Zn levels in Typha latifolia roots reached up to 1400 mg/kg. After contact with culture (initial sediment) for month one (without medium concentration of Zn was 1-10 mg/l), the Zn levels in Typha latifolia roots reached up to 3600 mg/kg (Table 2.4). Pinto et al. (1987) showed that after cultivation of water hyacinth plants for 24 hours in a 40 mg/l silver solution, the average concentration of silver recovered from the dried plant material was 70% of the initial silver concentration of the solution with a purity of 98%. This high uptake implies extremely high tolerance to specific elements as confirmed by Baker and Brooks (1989) in their study of the hyper-accumulation of metallic elements by plants.

These investigations would therefore imply a basic ability of macrophytes for indicating anthropogenic pollution and also in removing pollutants from wastewater.

treatment	Maximum content	Mean	Maximum
	per individual	values	values
	plants (g/kg)	(g/m ²)	(g/m ²)
Cd 1.0 ppm	0.146	1.46	6.56
4.0	0.393	3.93	17.7
8.0	0.124	1.24	5.60
Pb 1.0 ppm	1.05	10.5	47.5
4.0	7.46	74.6	336
8.0	13.9	139	627
Hg 0.5 ppm	0.007	0.77	3.47
1.0	2.34	23.4	105
2.0	0.17	1.70	7.52

Table 2.3: Maximum heavy metal removal values by water hyacinth (from Muramoto, 1983)

Zn Compound	Concentration	Main Roots	Plant Rhizome	Roots	New Plant Rhizome	Leaves
	mg/kg of tissue (dry wt)					
Zn Chloride	1 mg/l (A) 10 mg/l (B) Ratio (B)/(A)	650 2400 3.7	100 270 2.7	2500 3600 1.4	390 1800 4.6	68 250 3.7
Zn-EDTA	1 mg/l (C) 10 mg/l (D) Ratio (D)/(C)	310 1000 3.2	26 80 3.1	380 1500 3.9	200 1100 5.5	30 350 11.7

Table	2.4:	Comp	parative	increa	ses of	cond	centra	ations	of	Zn	in
	T	vpha	latifoli	ia (fro	m Blake	e et	al	1987)			

Harrison and Chirgawi's research (1989) of both soil and air as contributors of some trace metals to plants further and the efficiency of soil uptake that suggests atmospheric deposition are in the order of Zn>Cd>Ni>Cr>Pb (sometimes Cd>Zn), while Xian's (1989) study of the response of the kidney bean to concentration and chemical form of Cd, Zn and Pb in polluted soil shows that the solubility according to their was uptake of metals Cd>Zn>Pb. This latter hierarchy confirms the earlier work of Ellis et al., (1986) referred to in section 2.2.3.

2.2.6 The Effects of Heavy Metal Uptake on Macrophytes

Variation in the long-term metal uptake rate has been found to depend upon the species of plant, with higher heavy metal contents being associated with submerged species such as <u>Elodea</u> sp. and <u>Callitriche</u> sp. compared to floating species such as <u>Lemna</u> sp. and <u>Spirodela</u> sp.

(Werff and Pruyt, 1982) This suggests that high biomass species rooted in the sediment have a greater ability to remove pollutants than species exposed to the water column alone. Some existing studies (Merchyulenene and Nyanishkene, 1976; Leland and McNurny, 1984) strongly submerged macrophytes suggest that are capable of attaining tissue heavy metal concentrations far in excess of concentrations in the water column.

Plants appear to accumulate a higher concentration of metals in the roots than in the stems and leaves (Reimer, 1989). Welsh and Denny (1980) as well as Heisey and Damman (1982) have reported root to shoot ratios for lead concentrations of 3.8 - 5.8 and 3:1, respectively.

are Uptake rates also dependent upon changes in the seasonal growth rate. Actively growing plants of any species absorb metallic ions much faster than old or dying or seasonally decomposing parts of plants. For example, Zn levels in the leaves and stems of Phragmites australis and Cu concentrations in the stems are at a maximum during the and decrease thereafter (Larsen and growing season Schierup, 1981). Concentrations of Pb in the leaves have also been found to increase during and after the growing season (Fig. 2.2). The uptake rate is also dependent upon the specific metal ion being absorbed (Mortimer, 1985). The efficiency of uptake of metals from soil is generally thought to be high for Zn and Cd and low for Pb (Harrison and Chirgawi, 1989).



non-polluted water

o----o sewage-polluted water

Fig.2.2 Seasonal variation in concentrations of Zn, Cu and Pb in leaves (a,c,e) and stems (b,d) of <u>Phragmites australis</u> (from Larsen and Schierup, 1981).

There are some fairly obvious factors which are known to affect the efficiency of metal uptake including root mass, pH, chelating agent concentration and the presence of other competing metals. Hardy and Raber's study (1985) of Cd uptake by water hyacinth demonstrate that plants with greater root mass remove Cd more efficiently from solution per unit time. This study also shows that there is a sharp increase in the initial rapid uptake at a pH of about 5. Nitrilotriacetic (NTA), Ν -(2 hydroxyethy1) ethylenedinitrilo - N, N', N' - triacetic acid (HEDTA), ethyenedintrilo - N, N, N', N' - tetraacetic acid (EDTA) and trans - 1, 2 - cyclohexylenedinitrilotetraacetic acid (CDTA) significantly reduced the rate of Cd and Zn uptake by water hyacinth (Hardy, 1985). There are six metals which extensively hindered uptake of the Zn. These are Fe^{3+} , Cu^{2+} , Hg^{2+} , Co^{2+} , Cd^{2+} and Ni^{2+} .

Since ammonium ions, metal ions, chelating agents and pH levels can vary considerably in polluted water, the condition of the water with reference to these parameters must also be taken into consideration when macrophytes are to be used as an indicator of water pollution or to purify wastewater (Nasu and Kngimoto, 1983; Hardy and Raber, 1985).

2.2.7 The Mechanisms of Heavy Metals Uptake

River plants take up metals from both sediment and water, sediment usually being the more important source (Haslam, 1987). Emergent plants on the bank eg. <u>Phragmites communis</u> obtain all their minerals from the sediment. They absorb little minerals through their thick rhizome or aerial parts. So effective mineral uptake occurs only when living roots are present (Haslam, 1987).

- 20 -

Heavy metals can be taken up either from a sediment (soil)-root pathway or foliar route. On the one hand, heavy metals are taken up from sediments by roots and translocated to the rhizomes and aerial parts of emergent species. The greater part of heavy metals taken up by the plants would appear to be retained in the roots and rhizomes, with only small amounts being translocated to the above-ground part of the plant. For instance, up to 77 to 97% of heavy metals taken up by <u>Phragmites communis</u> is retained in the roots and rhizomes according to Larsen and Schierup (1981) and about 80% of the total cadmium uptake by water hyacinth after 30 days exposure was found in the roots by Blake et al. (1987).

However, analysis of the metals in the plant tissues (Harrison and Chirgawi, 1989) showed that the foliar route is potentially of similar importance to the soil-root pathway as a route of transport to the exposed parts of the plants. Metals entering the plant from the atmosphere can be translocated to unexposed portions of some plants, eg. radish and turnip. The exposed parts showed the highest metal accumulation with lower values found for indicative of translocation unexposed plant parts, of atmospherically-derived metal within the plant by for soil-derived mechanisms similar to those metal. Experiments with rainwater show a generally positive wet deposition component, rather than the contribution of dry deposition to the metal content of the plant.

Plants can grow on toxic sediment or soil: either they avoid their uptake and restrict the transport of the metal, or they have an internal mechanism that renders the metal harmless (Baker and Walker, 1990). Tolerant plants have high concentrations of metals in their tissues, eg. Eichhornia crassipes (Chigbo et al., 1982) and Typha latifolia (Blake et al., 1987). Clearly, they must have evolved a way to protect their life processes. Plant physiologists have proposed a number of mechanisms (Farago et al., 1975; Kelly et al., 1975; Kersten et al., 1980; Baker and Walker, 1990). One suggestion is that once the metal enters plant cells, compounds that bind to the metal the cytoplasm are produced. Once bound, the metal in complex may stay in the cytoplasm or move to another part of the cell, preventing further contact with metabolic processes. In either case, the toxicity of the metal is greatly reduced. Compounds suggested for the binding role in the vacuole include organic acids and sugar phosphates; in the cytoplasm the probable binders are phytochelatins (Thurman and Hardwick, 1988).

Zinc taken up by plants from soil and sediment is in either free ionic form or translocated as simple organic acid complexes to the shoots. Accumulation of Pb by the foliage of plants is an exceedingly rare event due this the readiness with which element can to be precipitated as the insoluble sulphate in the rhizosphere, hence minimalising potential uptake and transport to the aerial parts of the plants (Baker and Brooks, 1989).

2.2.8 Heavy Metal Removal

The partial removal of heavy metal pollutants by macrophytes can be achieved by frequent harvesting of the foliage and stems where a proportion of the accumulated metals are stored in the plant biomass. The potential rate of pollutant storage by an aquatic plant is limited by the growth rate and plant biomass per unit area as well as by uptake rate. The ease of harvesting macrophytes may be of
some significance in terms of their use as a wastewater purification technique, though this is probably less important than the capacity of the whole system to act as a sink. Muramoto and Oki (1983) reported that the maximum values of metal removed by water hyacinth are: 177 kg/ha from water containing 4.0 mg/l Cd, 6,270 kg/ha from water containing 8.0 mg/l Pb, and 1,050 kg/ha from water containing 1.0 mg/l Hg (Table 2.3).

If plants are not harvested, heavy metals can be released to the surrounding aquatic environment by plant decomposition. Black et al.'s studies (1979, 1984) showed that 25% to 42% Zn accumulated in Typha latifolia leaves and potentially could be released from the decomposed leaves. It is therefore necessary to cut and harvest the plant, as this technique allows the export of 25 % to 42 % of the Zn accumulated in Typha latifolia and prevents the leaching of the metal from decaying organic matter. DeBusk and Reddy's study (1990) shows that after injecting Pb (900 mg with concentration of 1.0 mg/l) and Cu (2250 mg with concentration of 2.5 mg/1) continuously for two into raceways containing pennywort (Hydrocotyle months umbellata), approximately 69% of the 14.0 g/m^2 Cu and 85% of the 4.9 g/m^2 Pb applied were removed with effluent concentrations averages 839 ug Cu/l and 149 ug Pb/l.

2.2.8 Heavy Metal Release

In order to understand the heavy metal cycle in an aquatic environment, it is very important to have a knowledge of the decomposition process. Several reports have been published on the behaviour of heavy metals in the decomposing litter contained within aquatic environments. Larsen and Schierup (1981) found that heavy metals tend to show an increase in concentration in decomposing <u>Phragmites</u> sp. litter during decomposition. From a one year study in a non-polluted lake, concentrations of Zn, Cu, Pb and Cd (mg/kg oven dry weight) in <u>Phragmites</u> leaves increased from 108.8, 5.0, 5.5 and <0.05 to 135.2, 17.0, 13.6 and 0.56, respectively.

By comparing the concentration of the different heavy metals in litter during decomposition to the concentration of the plant tissues prior to decomposition, heavy metals were observed to behave in different ways. The observed increase in metal concentration may arise as a result of a greater decrease in biomass compared to metal weight during decomposition and therefore does not necessarily relate to an actual increase in total metal load. During the decomposition of Phragmites sp. the relative contents of Zn and Cu in decomposing litter were either similar to that prior to decomposition or significantly lower. This might be a consequence of the leaching of Zn and Cu being easier under the specific aquatic conditions being studied than for Cd and Pb. The relative contents of Pb and Cd in partly decomposed Phragmites sp. leaves were however, prior significantly above those to found to be decomposition. The accumulated Pb was presumably absorbed from the surrounding water column or derived from a reduction the metal following a in concentration of biomass.

2.2.10 Biological Accumulation of Heavy Metals in the Marine Environment

Industrial and demographic growth along coastlines poses a threat, principally to estuaries, which are primary recipients of pollutants and critical intermediates in the biological cycle of marine organisms (Seeliger and Lacerda, 1986). The biological accumulation of heavy metals in marine environments has become a widely accepted concept. Heavy metal concentrations in benthic marine algae are magnified several thousand times with respect to their concentrations in seawater (Table 2.5). Due to their pollution tolerance, macroalgae have been considered the most suitable group of organisms for biological metal monitoring in coastal waters (Seegier and Cordazzo, 1982).

Metal	Blidingia	Algal g Enteromorpha	jenera Ulva	Fucus	Source
Cu	0 . 71 - 1.83	0 . 56-0.77	0.47-0.86 0.12	0.36-0.74 0.25 0.45 0.45-2.7	Seeliger(1977) Yamamoto(1972) Black(1952) Preston(1972) Bryan(1973)
Pb	2.7-8.2	2.0-4.5	1.7-4.9	1.3-2.4 0.087 0.23 0.24	Seeliger(1977) Black(1952) Preston(1972) Bryan(1973)

Table 2.5: Concentration factors $(x \ 10^4)$ in benthic algae tissue (ug/g dry wt) and dissolved metals in water (ug/ml)

Metal accumulation by algae is, at least in its first stage, a passive physico-chemical procedure occurring on the surface of algal cells. The procedure involves the binding of metal ions within the cell wall matrix, which contains anionic groups, with large cation exchange capacity (Reed and Darrig, 1983). The large biomass of

- 25 -

benthic algae in coastal waters will eventually undergo decomposition ultimately be and mineralized (Riley , 1970). Traditionally, in studies of biological magnification the primary producers have been considered to be consumed directly by herbivores, and thus their metal complement passed directly into the next link in the food web. However, in most coastal waters the benthic algae are largely unconsumed and their organic matter is passed along the food web as detritus and dissolved organic matter (Seeliger, 1979). It has been suggested that heavy metal concentrations in seaweeds are not subject to short term erratic fluctuations as are metals in water, but integrate the metal concentrations in water over relatively long intervals of time. This makes them valuable tools for the monitoring of heavy metal levels in the ambient water over extended periods of time (Seeliger, 1979).

2.3 Nutrient Pollution Removal

Nutrient pollution has been the source of a number of environmental problems, even though dissolved mineral calcium, nitrogen, phosphorus and such as nutrients for the maintenance of the essential are potassium production of aquatic plants and animals. The overload of nutrients to receiving waters has caused a rise in the primary plant productivity which has resulted in а reduction of the oxygen in the water and the death of fish. The minerals are taken up from both the water and soil and accumulate in roots, stems and leaves (Haslam, 1987). As a result macrophytes participate actively in the nutrients, retaining of and removal biological accumulating them in aquatic environments. It is now firmly established that macrophytes remove many biogenic elements and act as accumulators of nutrients (Kovalskii and Grivovskaya, 1976; Sundblag and Wittgren, 1989).

2.3.1 The Effects of Nutrient Uptake

Tsutomu and Teranish (1988) showed that the total N uptake correlated with rate was the Ν concentration by a Michaelis-Menten type equation, in which the maximum uptake rate of N was expressed as a function of the temperature and plant density. The P uptake rate however was influenced by not only the Ρ concentration, temperature and plant density, but also by variation in the P content of the plant. The accumulating ability of macrophytes shows species variation and depends on the growth location and season (Morozov, 1984). The removal of biogenic substances from water by macrophytes, which is always at a maximum rate at the start of the growing season, decreases by the end of the growing season as a result of losses due to excretion from plant tissues back environment. For example, the common the reed to <u>Phragmites communis</u> with a dry mass of up to 3 kg/m^2 during growth, can extract about 45g nitrogen and 18g phosphorus from water per annum. Narrow-leaved reedmace Typha angustifolia with a dry mass of 3.6 kg/m² can extract 32 g phosphorus per annum (Orekhovskii, 1986). Laboratory experiments with wastewater have shown that and water-weed. Elodea the narrow-leaved reedmace can extract up to 60% of the ammoniacal canadensis, nitrogen, more than 80% of the nitrates, up to 99.98% of nitrites and 37 - 57% of the phosphorus over a period of 1 week (Morozov, 1984). Water hyacinth Eichhornia crassipes can remove 71.3% ammoniacal nitrogen, 50% phosphorus and 30% potassium after 5 days (Xie et al., 1984).

2.3.2 Nutrient Storage and Release

Studies have shown that of more than 50% nutrients extracted by macrophytes were stored in below ground portions of the plants (Reddy and DeBusk, 1987). Emergent macrophytes have more supportive tissues than floating ones and so they have greater potential for storing the nutrients over a longer period. Typha latifolia can absorb 600 - 2630 kg/ha/yr of nitrogen and 75 - 403 kg/ha/yr of phosphorus. In addition, 250 - 1560 kg/ha and 45 - 375 kg/ha nitrogen and phosphorus, respectively, can be stored (Table 2.6).

			· · · ·		
Plant	Ni	Nitrogen		Phosphorus	
	Storage	Uptake	Storage	Uptake	
FLOATING MACROPHYTES:	kg/ha	kg/ha/yr	kg/ha	kg/ha/yr	
Eichhornia crassipes	300-900	1950-5850	60-180	350 - 1125	
Pistia stratiotes Hydrocotyle	90-250	1350-5110	20 - 57	300-1100	
<u>umbellata</u> Alternanthera	90-300	540-3200	23 - 75	130-770	
philoxeroides Lemna minor	240-425 4-50	1400-4500 350-1200	30-53 1-16	175 - 570 116 - 400	
Salvinia rotundifolia	15-90	350-1700	4-24	92-450	
Typha spp.	250 - 1560 200 - 300	600 - 2630 800	45 - 375 40	75 - 403 110	
<u>Scirpus</u> Phragmites	175-530 140-430	125 225	40 -11 0 14 - 53	18 35	

Table 2.6: Storage of nitrogen and phosphorus and rate of plant uptake for selected aquatic macrophytes (from Reddy and DeBusk, 1987)

Plant species which have a higher relative biomass can store higher nutrient load. Thus, in order to improve water quality, the most efficient way to remove nutrients is biomass to remove the by harvesting the plants (Frederiksen, 1987). If plants are not harvested, the dead tissue will decompose rapidly and release nutrients into the water. Frequent harvesting of biomass is therefore essential and necessary to avoid losses of nutrients back to the water column. Macrophytes essentially extract all of the nutrients they require for their growth entirely from the benthic sediment and release the nutrients during decomposition back to the water. So the life cycle of macrophytes may provide an important source of nutrients to the aquatic environment. When a high biomass exists, macrophytes can play an important part in nutrient cycles of aquatic ecosystems (Jewell, 1971; Howard-Williams and Davies, 1979; Carpenter 1980, 1981; Werner and Adams, 1984).

2.4 Hydrocarbon Pollutant Removal

2.4.1 Hydrocarbon Degradation

is achieved through bacterial degradation Hydrocarbon activity. Oxygen is therefore a very important factor as affects microbial growth. Aquatic plants have the it unique feature of transporting 02 through the leaves, sediment Transported oxygen enters and roots. stems utilised by aerobic water, and is interstitial bacteria during the oxidation of hydrocarbons (Foght and Westlake, 1987). Oxygen transferred by plants into the root zone therefore plays a significant role in supporting the aerobic bacteria in this area and in the degradation of hydrocarbons. For example, 0.02 - 0.1 g of the root mass of <u>Typha latifolia</u> can transport 1.39 ± 1.49 mg $0_2/g/hr$ and 0.03 - 0.06 g of the root mass of <u>Sagittaria</u> <u>latifolia</u> can transport 1.72 ± 0.87 mg $0_2/g/hr$ (Table 2.7).

Plant	Root mass	0 ₂ transport	n
	g	mg 0 ₂ /g/hr	
Hydrocotyle umbellata	0.02-0.05 0.06-0.12	3.95±1.86 2.49±1.05	18 8
<u>Pistia stratiotes</u>	0.05-0.25	0.30±0.13	10
Eichhornia crassipes	0.03-0.10 0.11-1.25 0.26-0.50 0.51-0.99	1.29±1.18 1.27±0.61 0.31±0.11 0.12±0.14	10 10 8 4
Sagittaria latifolia	0.03-0.06	1.72±0.87	15
Typha spp.	0.07-0.14 0.02-0.10 0.11-0.53	0.61 ± 0.22 1.39±1.49 0.19±0.15	3 4 14

Table 2.7: Oxygen transport through selected aquatic macrophytes (from Moorhead and Reddy, 1987)

2.4.2 Oil Clean-Up by Plants

The reported biological effects of repeated small oil spillages and chronic discharges range from localized and subtle to widespread and persistent (Dicks and Hartley, 1982). The vascular plants most commonly affected by oil pollution are saltmarsh and mangrove species (Baker, 1983). The study of oil spill clean-up techniques in <u>Spartina alterniflora</u> (Kiesling et al., 1988) shows that clipping followed by sorbent pad application on the - 30 - substrate removed 36 - 44% of added oil. Pilot plant trials with palm oil mill effluent have shown that water hyacinth can also be successfully grown in anaerobically digested liquor removing 97% oil and grease (John, 1984). Delaune et al. (1984) found a 31% reduction in sediment oil after clipping. In littoral areas, oiled substrate stripping is viewed as a radical clean-up approach.

2.5 The Application of Aquatic Macrophytes as Wastewater Treatment Systems

of their Because ability to remove pollutants from wastewater and the attendant low capital cost, many biotechnological wastewater treatment systems have been used to improve water quality such as those offered by high rate algal ponds (Shelef and Soeder, 1980), duckweed systems (Oron and Wildschut, 1984), water hyacinth systems Tchobarioglous, 1985), (Weber and land application 1985), surface-flow (Abernathy et al., reed beds artificial and 1985) wetland (Alexander, wastewater treatment systems (Seidel, 1976; Kickuth and Kaitzis, 1975). Among these alternatives, the root zone treatment systems and water hyacinth treatment systems are probably the most popular.

2.5.1 Use of Water Hyacinth in Wastewater Treatment

Of the known vascular aquatic plants, one of the most promising for industrial utilization is water hyacinth (<u>Eichhornia crassipes</u>) which, in view of its special characteristics, is an excellent material for a biological filtration system. The water hyacinth has been widely used for wastewater treatment in many countries because of its prolific growth rate and ability to remove pollutants from wastewater. It was noted that two plants produce 300 units in 23 days and can cover an area of 4,000 km^2 in 8 months (Wolverton and Mcdonald, 1979). Biomass production is around 873 kg/ha/day matter (Wolverton and McDonald, 1975). Water hyacinth treatment systems can reduce COD, TSS, N, P, K, heavy metals, pesticides, petrol and oil (Gupta, 1980). John (1984) showed that water hyacinth can be successfully grown in anaerobically digested liquor removing 96% BOD, 87% COD, 96% suspended solids, 83% ammoniacal nitrogen and 97% oil and grease after 25 - 30 2.8). days (Table It also removed more than 99% of indicative bacteria such as coliforms, Escherichia coli and Streptococci. Widyanto et al. (1983) suggested that also absorb sodium, chlorine water hyacinth can and sulphur from the wastewater. Xie et al.(1984) showed that the concentrations of Cu and Zn in sewage water decreased notably after absorption by water hyacinth for 15 days. Hardy and O'Keeffe (1985) showed that after exposure to Cd^{2+} for 4 hours, 300 mg dry weight rootmass can remove about 400 ug. Studies have shown that water hyacinth is suitable for treatment of industrial wastewater such as textile wastewater (Trivedy and Gudekar, 1987); natural rubber effluents and oil palm mill effluents (John, 1984) as well as pisciculture waters (Simeon and Silhol, 1987) and domestic sewage (Cloris and Araujo, 1987).

Santos et al. (1987) showed that, on a qualitative basis, the following assertions can be made to remove BOD:

a) Without water hyacinths in the reactor, the effluent would have a poor appearance due to the algae which would influence the BOD if not filtered. Even after filtering, the BOD could still increase due to the extracellular products usually excreted by algal metabolism. b) The fixation of micro-organisms on the plant roots of water hyacinths increased the quantity of active microorganisms inside the water.

c) It was observed by microscopic examination that the micro-organisms present on the plant roots were strictly aerobic. This aerobic zone could only be explained by the presence of the hyacinths. Once the aerobic zone is formed, there is an obvious increase in the decomposition rate due to aerobic metabolism, and а valuable contribution from the microfauna present on the roots (protozoa, rotifers, etc.) in the removal of carbonaceous matter.

Parameter*	Raw	Hydraul	Hydraulic retention time (day)				
	effluent	10	20	25			
pH BOD COD Suspended solids Total solids Total nitrogen Ammoniacal nitrogen Oil and grease	6.49 4980 8850 3560 8210 285 120 660	7.49 660 2060 550 5550 140 55 25	7.98 310 1500 450 4690 80 25 30	8.08(-) 180(96.4) 120(87.3) 140(96.1) 4540(44.7) 65(77.2) 20(83.3) 20(97.0)			

Table 2.8: Effect of Growing Water Hyacinth In Partially Digested Liquor (from John, 1984)

*All parameters except pH are expressed in mg/l. The figures in brackets refer to percentage reduction. 2.5.2 The Root Zone Biotechnology System of Wastewater Treatment

Emergent plants such as the common reed (Phragmites communis), common reedmace (Typha latifolia), narrow leaf reedmace (Typha angustifolia), bullrush (Schoenoplectus lacustris) etc. can grow rapidly in urban wetlands, are pollution tolerant, and produce a high biomass. In addition they can actively remove pollutants from wastewaters and supply oxygen to associated bacteria which proliferate in the aerobic conditions, thus creating stabilized organics in the effluent and nitrifying ammonia to nitrate (Lawson, 1985; Alexander and Wood, 1987).

The macrophyte bed treatment systems approaches have been pioneered by both Seidel (1976) and Kickuth (1975). Seidel's process, known as the Krefeld system, utilizes Typha spp. planted in a gravel substratum (Fig. 2.3 a). Kickuth's process, known as the root zone system, utilizes Phragmites spp. planted in a soil substratum (Fig. 2.3 b). Both systems were developed for sewage treatment purposes the removal of organics and ammonia from the with wastewater as the prime consideration. Further variations on helophyte bed systems are shown in Fig. 2.3 c - e. The Biotechnology (RZB) reed-bed Root Zone systems and (Arthur, 1986) however have now become well established procedures in continental Europe and there are currently forty three Reed Bed Treatment Systems in operation at twenty seven sites within the UK.



(a) Helophyte beds in series with gravel or sand as suggested by Seidel (1966, 1983)



(b) Helophyte bed with cohesive soil as suggested by Kickuth (1980, 1984)



(c) Helophyte bed with gravel or sand, alternative solution to (a)



(d) Helophyte bed with cohesive soil, alternative solution to (b)



(e) Helophyte bed with cohesive soil, alternative solution to (b)



It is an attractive technology, offering relatively low capital and revenue costs to attain water of a given quality (Loehr et al., 1979; Overcash and Pal, 1976) and which can cope with many types of effluent over and beyond food and domestic wastes (USEPA, 1977). RZB is ideal for small-scale/low-technology treatment situations, e.g. farm wastes. landfill leachate or treatment systems in developing countries. In the UK and Europe the use of reed beds as а controlled wetland for treating domestic effluent is currently attracting considerable interest. Studies have shown that aquatic macrophyte-based treatment systems offer a low-cost method of removing 80% to 90% of organic loading (BOD₅, COD) and 85% TSS (Suspended Solids) from domestic wastewater (Boutin, 1987) and can remove 76% of COD and 82% of total-sulphur load from textile finishing effluent (Winter and Kickuth, 1989). Reed-beds can remove about 85% of nitrates and a large amount of phosphates as well as hydrocarbons, phenols, heavy metals and pathogenic bacteria. Reeds have the remarkable ability of absorbing oxygen through their above ground stomatal pores and of transporting it down to the root zone. This multiplies the oxidation value of the boundary surface intensity of chemical 40 times, hence the and some bacteriological processes in the root zone. The root zone treatment systems have been used for (1) raw sewage, (2) settled sewage and (3) biological treated sewage treatment (Bucksteeg, 1987). The pollutant removal capability of RZB systems has therefore been widely proven in wastewater sewage treatment particularly to meet effluents and objectives at low cost in small treatment tertiary communities (Cooper et al., 1989).

Despite these encouraging developments, the technology has its limitations and is certainly not a general panacea for urban waste-water treatment. Consideration must be given to (a) the problems of public acceptance, (b) the potential contamination of ground and surface waters, (c) pathogens, and (d) contamination of the food chain (Bayes et al.,1989) as well as the potential volumes and loading rates required to be handled in urban areas.

A number of useful applications for water hyacinths have been investigated (Bates and Hentges, 1976; Widyanto et al., 1983; and Abdalla et al., 1987). These include schemes for harvesting to make compost and soil additives, to extract chlorophyll and carotene, to produce highprotein cattle food, to produce pulp, paper and fibre and more importantly to provide biogas such as methane as an energy source. The percent of methane in the biomass produced by anaerobic decomposition of water hyacinth contaminated with heavy metals is 91.1% while the noncontaminated plants produce biomass with 69.2% methane. One kilo dry matter produces 140 - 280 litre methane in 23 days (Wolverton and McDonald, 1975). The residual sludge may be utilized for recycling the metals (Table 2.9).

Table 2.9: Processes for the transformation of vascularaquatic plants into industrial products(from Pinto et al., 1987)

Absorption and Removal of Substances	Plant Processes		Products
Removal of Heavy Metals and organic	Anaerobic Fermentation		Methane
Substances from Industrial Effluents	Residual Sludge	- Metal Extraction	Ag, Au, Cd, Hg, Pb, etc
	Aerobic Fermentation Residual	Drying	Methane Fertilizers
Removal of		Animal Food	Added to ration of cattle pigs, etc
Nitrates from Domestic Sewage	Dry Matter Processing -	 Human Food	- Protein Supplementation (flour or bran)
	Composted		Fertilizer for Agriculture

Chapter 3: Locations and Methods

3.1 Field Study

3.1.1 Sampling Locations

Individual (<u>Typha latifolia</u>) plants and associated sediment and water samples for heavy metal analyses were collected at approximately bimonthly intervals between October 1988 and December 1989 from three sites in N London located within the London Borough of Barnet and having differing background levels of pollution (Fig. 3.1). These included a spring-fed pond situated in a country park estate (site 1), an ornamental pond receiving surface water discharges from the M1 motorway (site 2), and the Welsh Harp flood storage reservoir (site 3) on the lower reach of the Silk Stream feeder to the River Brent.

Site 1 (grid reference TQ 293973): This spring-fed pond, the Dew Pond, is situated within the Middlesex Polytechnic campus of Trent Park, on the 'green belt' fringe of North London (Fig. 3.2). The Dew Pond has an extensive bed of Typha latifolia whose annual leaf decomposition has contributed organic material to the benthal sediment. The absence of visitor pressure and management has led to the dense vegetation, including Salix sp, Crataegus sp and Rubus sp around the pond (Shutes, 1985). The pond receives some surface runoff from adjacent arable fields and meadows resulting in limited pollutant inputs. Samples were collected from the east margin of the pond (Plate 3.1).





- 40 -





- 41 -



Plate 3.1 Site 1 The Dew Pond situated in Trent Park

Site 2 (grid reference TQ 203931): This ornamental pond in Stonyfield Park is fed by the Dean's Brook a tributary of the Silk Stream, which receives surface runoff from the M1 motorway and other local road surfaces in NW London (Fig. 3.3). The Dean's Brook is one of the two tributaries of the Silk Stream and rises from a series of springs above a nearby golf course.

Samples were collected at the north west margin of the pond where the fringing vegetation is dominated by <u>Typha</u> <u>latifolia</u> (Plate 3.2).

- 42 -







Site 3 (grid reference TQ 216882): The northern arm of the Welsh Harp reservoir in NW London comprised the third sampling location. This basin receives urban stormwater runoff via the Silk Stream which enters at its northern end (Fig 3.3). The catchment of the Welsh Harp is dominated by stormwater inputs from both combined and separate sewer systems, with 60% of the storm flow being derived from impermeable surface runoff (Hall, 1977). The Welsh Harp was built in the mid-nineteenth century by constructing a dam across the river Brent which rests on a bed of London Clay to create a total impoundment area of 190 acres. Until the early 1920s, the drainage basin but subsequent urbanisation has mainly rural was increased the percentage of impervious area in the catchment from 15% in 1929/30 to 25% in 1975/76 (Gavens et al., 1982). Urbanisation has increased at a rate of around one per cent of the land area each year and now comprises some 43% of the catchment. The reservoir is now a designated Site of Special Scientific Interest (SSSI), of bird species and the marginal its on account successional vegetation formed on the shallow areas of the Northern and Eastern Arms. There is an extensive area crack willow (Salix dominated by of woodland wet fragilis) surrounded by successional vegetation composed of greater and lesser reedmace (Typha latifolia and Typha angustifolia).

Samples were collected at the site which receives the Silk Stream at northern edge (Plate 3.3).

- 44 -



Plate 3.2 Site 2 The ornamental pond situated in Stonyfield Park



Plate 3.3 Site 3 The northern arm of the Welsh Harp reservoir

3.1.2 Sampling Methods

Large plastic containers, plastic bags and plastic sample bottles pre-washed with 2% nitric acid were prepared for plant, sediment and water samples respectively. Plants were dug out by spade and kept in the plastic containers. Surface sediments (top 2.5 cm) were obtained by grab samplers and stored in plastic bags. Water samples were collected in plastic bottles which were pre-washed with sample water and then filled to 2/3 capacity.

3.1.3 Laboratory Methods

Plants: On return to the laboratory, the plant samples were carefully washed to prevent tissue damage with tap water to remove any foreign material attached to the surface (leaves were washed with 2% diluted nitric acid) followed by deionised water to remove all traces of attached sediment. The plants were separated into three parts constituting leaf, rhizome and roots and each fraction was oven dried for 24h at 100°C. After grinding stainless steel mill, the oven dried plant with а digested for 24h with 10-15 m1 material (2.5g) was concentrated nitric acid in a 100 ml pyrex beaker placed on a heated sand-bath and taken to dryness. The resulting residue was taken into solution with 2% hydrochloric acid and after filtering through a Whatman No.42 filter paper, the solution was made up to 100ml with deionised water. Duplicate samples were prepared for all plant fractions.

<u>Sediments</u>: Sediment samples were dried for 24h at 100°C in a hot-air oven and the dried sediment material (2.5g) was digested to dryness with 10-15ml concentrated nitric acid in a 100ml pyrex beaker placed on a heated sand-

- 46 -

bath. The resulting residue was taken into solution with 2% hydrochloric acid and after filtering through a Whatman No.42 filter paper, the solution was made up to 100ml with deionised water. Duplicate samples were prepared for all sediment fractions.

<u>Analytical Techniques for Plants and Soil</u>: The concentrations of Cu, Pb, Zn and Cd in the extracted samples were measured by means of a Philips SP9 Atomic Absorption Spectrophotometer (AAS) in the flame atomisation mode (Plate 3.4).

Standards were prepared from Spectrosol standard solutions (1000 ppm). A 100 ppm stock was used to prepare standards of 1, 2, 5, 10 ppm of Cu, Pb, Zn, Cd for calibration. The SP9 Computer was set to a fixed ratio programme of 4 standards giving automatic calibration with curvature correction.

The AAS was used with the following operating conditions:-

(i) deuterium B/C correction

(ii) 0.2 nm bandpass setting

(iii) air/acetylene flame

(iv) 10 sec. integration time

Each lamp was allowed an initial heating up time of 10 minutes and a little less than maximum current was used in order to give greater sensitivity. Cu, Pb, Zn, and Cd lamps were set at wavelengths of 342.8 nm, 217.0 nm, 213.9 nm and 228.8 nm respectively.

A blank of deionised water was used before the standards were measured in ascending order of concentration.

- 47 -

Once the system was calibrated, sample concentrations were measured and converted to parts per million (ppm) expressed on a dry weight basis as plant or sediment.

Water: Water samples (50ml) were heated to dryness on a hot sand-bath after the addition of 2ml concentrated nitric acid. The residue was taken into solution with 2% dilute nitric acid and after filtration, deionised water added to give a final volume of 50ml. was Duplicate samples were prepared for each water sample. The concentrations of heavy metals (Cu, Pb, Zn and Cd) in the water samples were measured by means of a Princeton Applied Research Model 384 Polarographic Analyser (Plate 3.5) using the standard addition method.

<u>Analytical Techniques for Water:</u> The operating procedure using the instrument parameters shown in Table 3.1 were follows:

(1) 7ml of the acid digested water sample and 3ml of sodium acetate buffer (2M) were placed in the sample cell and analysed by the Anodic Stripping Voltammetry (ASV) technique.

Parameters			
Initial E	-1.200 V	Equilibriation time	30 seconds
Final E	0.000 V	Pulse height	0.050 V
Purge time	240 seconds	Replications	1
Drop time	1.0 seconds	Deposition time	60 second
Scan increment	2 MV	Conditioning time	0 seconds

Table 3.1: Parameters used for Model 384 Polarographic Analyser

- 48 -



Plate 3.4 Philips SP9 Atomic Absorption Spectrophotometer used for sediment and plant analysis



Plate 3.5 Model 384 Polarographic Analyser used for water analysis

(2) The above solution was spiked with 0.02 ml of a mixed 10 ppm Cu, Pb, Zn and Cd solution and the ASV analysis repeated.

(3) Procedure (2) was repeated.

The results were printed as graphs and after the calculation the concentration was expressed as parts per billion (ppb).

3.1.4 Biomass Measurement

The <u>Typha latifolia</u> plants in a 0.5 m square quadrat were counted and completely removed from site 1 on 3 October, 1989. The numbers of shoots within each quadrat were counted. Plants were taken back to the laboratory and washed with tap water. The roots, rhizome and leaves of the plants were separated and their wet weights determined. The plant tissues were oven dried for 24 hours and their dry weights measured.

3.2 Greenhouse Experiments

Greenhouse experiments with <u>Typha latifolia</u> and <u>Juncus</u> <u>effusus</u> were carried out at the Middlesex Polytechnic Centre for Urban Pollution Research at Enfield and treated as follows.

3.2.1 Short Term Heavy Metal Dosing Experiments

<u>Typha</u> plants were collected on five occasions from the relatively unpolluted Dew Pond, at Trent Park (site 1) between 19 January, 1989 and 25 April, 1989. The plants were placed in 3 plastic tanks (35 cm high and 30 cm diameter) containing Irish Moss Peat. The background concentrations of heavy metals (Cu, Pb, Zn and Cd) in the peat were determined so as to assess any bias from contaminated peat. Irish Moss Peat was chosen because of its presumed low background metal levels (Cu, 2.0; Pb, 11.2; Zn, 19.2 and Cd,<0.01 mg/kg). 2 dosage tanks and 1 control tank were used, each containing three to five plants.

The plants dosed with were 2 mg/land 5 mg/1concentrations of Cu, Pb, Zn and Cd as their nitrate litres of the appropriate standard salts using 7.5 solution. The same amount of tap water was added to the control tank. After 10 days, plant, peat and water samples were collected from the 3 tanks and were analysed for metals by the methods outlined above. The leaf, rhizome and root tissues of Typha latifolia were analysed separately.

3.2.2 Long Term Heavy Metal Dosing Experiments

Long term heavy metal dosing experiments were also carried out over a period of eight weeks with both <u>Typha</u> and Juncus plants.

Dosing experiment with Typha latifolia: This experiment was carried out between 26 June, 1989 and 21 August 1989 using plants collected from site 1. Four dosage tanks and 1 control tank were used, each containing three to five plants.

Plants were dosed with 10 mg/l concentrations of Cu, Pb, Zn and Cd as their nitrate salts. 7.5 litres of 10 mg/l concentrations of mixed Cu, Pb, Zn and Cd solution was added to each of the dosage tanks while the same volume

,

of tap water was added to the control tank. One plant and associated peat samples were removed each week from the tanks and analysed by the dosage methods described previously (plant and associated peat samples from the control tank were analysed after 8 weeks). The peat samples were obtained from both the surface and bottom layers within the containers. The leaf, rhizome and root of Typha latifolia were analysed separately. The experiment was continued over 8 weeks and the plants in both dosage and control tanks were watered with beaker (water was added to sediment) approximately twice weekly (Table 3.2) with 10 mg/l concentrations of mixed Cu, Pb, Zn and Cd solution.

Date	Water [*] added (1)	Date	Water [*] added (1)
26/6 30/6 3/7 7/7 11/7 14/7 18/7 21/7	7.5 2.5 2.5 2.5 2.5 1.7 1.7 2.5	25/7 27/7 30/7 4/8 9/8 11/8 13/8 13/8 18/8	2.5 1.5 1.7 1.7 2.5 2.5 2.5 2.5 2.5

Table 3.2: Dosing Frequency and Water Volume Used for Dosing Experiment with Typha

* mixed 10 ppm Cu, Pb, Zn and Cd solution

Dosing experiment with Juncus effusus: This experiment was carried out between 3 October 1989 and 29 November 1989 using plants collected from site 1. Four dosage tanks and 1 control tank were used, each containing ten to fifteen plants.

Plants were dosed with 10 mg/l concentrations of Cu, Pb, Zn and Cd as their nitrate salts. 7.5 litres of 10 mg/l concentrations of mixed Cu, Pb, Zn and Cd solution was added to each of the dosage tanks while the same volume of tap water was added to the control tank. Five plants and associated peat samples were removed each week from the dosage tanks and analysed by the methods described previously (plant and associated peat samples from tank were analysed after 8 weeks). The control peat samples were obtained from both the surface and bottom layers within the containers. The leaf and root of Juncus were analysed separately. The experiment was continued over 8 weeks and the plants in both dosage and control tanks were watered approximately once weekly (Table 3.3) with 10 mg/l concentrations of mixed Cu, Pb, Zn and Cd solution.

Date	Water [*] added (1)	Date	Water [*] added (1)
3/10	7.5	31/10	1.7
10/10	2.5	7/11	1.7
15/10	2.5	14/11	2.5
20/10	1.7	22/11	2.5
26/10	2.5	29/11	2.5

Table 3.3: Dosing Frequency and Water Volume Used for Dosing Experiment with Juncus

* mixed 10 ppm Cu, Pb, Zn and Cd solution

Dosing experiment with peat: An additional experiment was carried out using a tank containing only Irish Moss Peat to assess the uptake metal capacity of the peat in the absence of plants. The tank was dosed approximately twice weekly (Table 3.4) with 10 mg/l concentrations of mixed Pb, Zn, Cu and Cd solution as their nitrate salts over 8 weeks period. The peat samples were obtained from both the surface and bottom layers within the container and analysed by the methods described previously.

Date	Water [*] added (1)	Date	Water [*] added (1)
18/1	7.5	17/2	2.5
23/1	2.5	20/2	2.5
26/1	5.0	24/2	2.5
2/2	2.5	30/2	5.0
8/2	2.5	5/3	2.5
13/2	5.0	10/3	2.5

Table 3.4: Dosing Frequency and Water Volume Used for Dosing Experiment with Peat

* mixed 10 ppm Cu, Pb, Zn and Cd solution

Chapter 4: Dosing Experiments

4.1 Introduction

Aquatic macrophytes are able to accumulate heavy metals to high levels (Chigbo et al., 1982; Muramoto and Oki, 1983; Sridhar, 1986; Blake et al., 1987), therefore, over the last 15 years these macrophytes have been studied extensively with regard to their use as indicators of pollution (Beeftink et al., 1982; Taylor and Crowder, 1983) and as water purifiers (Seidel et al., 1978; Black and DuBois, 1982). The emergent macrophyte Typha latifolia in particular grows rapidly and possesses a large capacity for assimilating mineral compounds (Black et al., 1984). A genetically based tolerance has been proposed for this species by Antonovics (1975). However, the ability of Typha populations originating from both contaminated and uncontaminated locations to tolerate high metal loads implies a constitutional tolerance (McNaughton et al., 1974; Taylor and Crowder, 1983). Typha has therefore been selected for studying metal uptake in this investigation. The aim of this study was thus to determine metal uptake by sediments and plants under greenhouse conditions by Typha latifolia. The metal uptake ability of Typha has been compared with Juncus effusus.

controlled greenhouse-based phytoassay experiments In Typha and Juncus were transplanted to Irish moss peat from set metal concentrations to dosed with and site 1 maximum and storage uptake rates determine both capabilities.

4.2 Discussion of the results

4.2.1 Short term dosing experiments

The short term greenhouse based metal uptake experiments were carried out over a ten day period (Chapter 3). <u>Typha</u> <u>latifolia</u> plants collected from site 1 were used for these tests and did not suffer any apparent toxic effects over the duration of the test. The results of metal uptake by peat and <u>Typha</u> plants over this time period are shown in Fig. 4.1. After 10 days, the metal levels in peat, <u>Typha</u> root, rhizome and leaf increased 72.1 %, 45.3 %, 36.7 % and 26.3 % respectively in 5 ppm dosed tests and when the dosing water concentrations increase, the concentration difference between peat, root, rhizome and leaf also increased.

Peat metal levels were considerably enhanced over the 10 day experimental period with this being most noticeable for 5 ppm dosed peat. Metal levels increased from a base of 0.5 to 34.8, 2.8 to 22.0, 12.2 to 27.2 and 17.8 to 37.6 mg/kg for Cd, Cu, Pb and Zn respectively. Cu and Cd levels in dosed peat exhibit much higher concentration ratio increases compared to Pb and Zn over the 10 day dosing period. The ratios of the final concentrations achieved by the 5 ppm dose relative to the control peat are 68.0:1, 7.9:1, 2.2:1 and 1.9:1 for Cd, Cu, Pb and Zn, respectively (Table 4.1). Cd levels in 2 ppm and 5 ppm dosed peat after 10 days experiment exceeded the Cd levels observed in plant root tissues, whereas Cu, Pb and Zn levels in the dosed peat remained lower than the metal levels achieved in the plant root tissue.







Fig.4.1 Heavy metal levels in peat and Typha tissues

Metal levels in the plant root, rhizome and leaf tissues increased consistently over the 10 days dosing experiment with the highest levels being demonstrated by the 5 ppm dosed plant root tissues. The highest concentrations of all the metals were found in plant root tissues with a progressive decrease to rhizome and leaf tissues.

		5 ppm dosed*	2 ppm dosed*	Control
				(mg/kg)
Peat	Cu Pb Zn Cd	7.9 2.2 1.9 68.0	5.6 1.7 1.4 48.0	1 (2.8) 1 (12.2) 1 (19.8) 1 (0.5)
Root	Cu Pb Zn Cd	2.3 1.4 1.5 2.6	1.4 1.3 1.2 1.4	$\begin{array}{c}1 (14.0)\\1 (30.4)\\1(117.0)\\1 (4.1)\end{array}$
Rhizome	Cu Pb Zn Cd	1.3 1.4 1.8 2.0	1.2 1.3 1.6 1.2	1 (8.1) 1 (10.6) 1 (30.4) 1 (3.7)
Leaf	Cu Pb Zn Cd	1.2 1.5 1.3 1.6	1.2 1.1 1.3 1.6	1 (7.6) 1 (6.3) 1 (26.0) 1 (3.3)

Table 4.1: Ratios of peat, root, rhizome and leaf metal levels relative to control peat

* dosing time 10 days

1
In parallel to the metal elevation in peat, Cu and Cd levels in 5 ppm dosed plant roots are more enhanced than Pb and Zn, whereas the ratios of dosed to control rhizome and leaf for all metals are relatively stable. A similar rapid uptake of Cd and Cu has been shown by experiments with <u>Callitriche platycarpa</u> and <u>Elodea nuttallii</u> over 14 days (Van der Werff and Pruyt, 1982).

4.2.2 Long term dosing experiments

In order to examine metal uptake by sediment and plant over the longer term, the dosing experiments were carried out over an extended eight week period of time with both <u>Typha latifolia</u> and <u>Juncus effusus</u>.

4.2.2.1 Long term dosing experiments with Typha latifolia

In the greenhouse-based metal uptake experiments carried out over an eight week period with Typha latifolia, the dosing metal levels of 10 mg/l were considerably enhanced in comparison to the average soluble metal concentrations found at the most polluted field site (site 3; Cd 8.9 ug/1: Cu, 53.4 ug/1: Pb, 36.2 ug/1 and Zn, 136.6 ug/1). These long term, high dosing level tests are not entirely with environmental field unrealistic comparison in the observed sediment metal situations where concentrations (site 3; Cd, 12.4; Cu, 220.1; Pb, 841.2 and Zn, 778.9 mg/kg) are considerably higher for Pb and Zn compared to those monitored in the Irish Moss peat during the duration of the experiment.

The surface peat metal levels demonstrated large enhancements over the eight week experimental period with linear increases from < 0.01 to 286, 2 to 187, 11.2

- 59 -

to 168.0 and 19.2 to 294.8 mg/kg for Cd, Cu, Pb and Zn respectively. The bottom peat metal levels remained relatively stable although with some random fluctuations (Fig.4.2). As the dosing time increases, the metal level difference between surface and bottom peat increases. The surface metal levels are up to 4.8 times higher than those of the basal peat. (Table 4.2). These results agree with a previous study by Blake et al. (1987) which showed a major proportion of added Zn was found in the surface sediment and the top 0.2 cm sediment layer had a concentration 2.5 times higher than the sediment Zn concentration found at a depth of 1.0 cm. In the present study, surface Zn concentration is 4.3 times higher than the bottom peat at a depth of 15 cm at the end of the 8 week experimental period (Table 4.2).

peat during experiment with Typha				
 	1-2	3-4	5-6	7-8 (dosing weeks)
 Cu Pb Zn Cd	2.0 1.1 1.7 1.2	3.1 1.5 3.0 1.7	3.5 2.3 2.4 1.8	4.8 3.4 4.3 4.1

Table 4.2: Metal concentration ratios of surface to bottompeat during experiment with Typha

,





Fig. 4.2 Metal level variations in surface and bottom peat samples during Typha latifolia 8 week dosing experiments

The surface sediment increases are most closely paralleled by the metal uptake patterns of <u>Typha</u> root tissue with the rhizome showing a reduced metal affinity and the leaf material being clearly the least important site with regard to metal concentrations (Fig. 4.3). Only in the case of Pb, does the leaf tissue eventually achieve comparable metal levels to the rhizome. This observation essentially agrees with that reported by Blake et al. (1987) in which the highest concentrations of Zn were found in the roots of <u>Typha latifolia</u> and were in the order of roots>rhizome>leaf tissue.

The observed peat, Typha root, rhizome and leaf metal uptake data over the eight week time period of the dosing analysed experiment were by statistical regression methods. The resulting equations are shown in Fig.4.4. The regression analysis shows that Typha leaf and rhizome metal uptake patterns are all linear (except for leaf Cd uptake which is exponential). The b values in the equations of leaf metal uptake are lower, so the patterns are more consistent. The root metal uptake patterns are consistent, with Zn and Cd both exhibiting less exponential increases with time and final values approach the These metals show greatest metal 700 mg/kg. enhancement in the root compared to the rhizome and leaf system but only after prolonged exposure.





Dosing concentration 10 ppm Dosing time 26/6/1989 - 21/8/1989

Fig.4.3 Temporal increases in peat and <u>Typha</u> tissue metal concentrations during 8 week dosing experiment



Fig.4.4 Regression equations and graphs for metal uptake by peat and <u>Typha</u> tissues during 8 week dosing experiments

- 64 -

The final root Cd and Zn levels are considerably elevated over those in the surface peat which corresponds to their established existence in urban waters as predominantly soluble bioavailable metal ions or weak complexes (Revitt and Morrison, 1987). However, the delay in uptake which can be seen from Fig.4.4 would appear to be due to the superimposed influence of sediment-water interactions which only release the bioavailable metals for plant uptake once a particular sediment saturation level has been reached. This saturation appears to be achieved after 4-5 weeks resulting in enhanced root uptake at this stage. Previous experiments with Typha latifolia (Blake et al, 1987) have shown Zn levels in the plant root tissue reaching 1400 mg/kg after dosing with 10 ppm solution for one month.

In comparison to Cd and Zn, Cu and Pb root tissue concentrations demonstrate lower accumulative metal levels after eight weeks reaching values of 190 and 242 mg/kg respectively. Cu root tissue concentrations exhibit a less exaggerated exponential increase with time whereas peat Cu concentrations achieve a value of 187 mg/kg. This is consistent with the established affinity of Cu with the humic and fulvic acids associated with soils (Luoma and Davis, 1983). In contrast to the other metals, Pb root tissue concentrations exhibit a linear increase showed less affinity of Pb compared to Zn and Cd. This is consistent with the previous studies of Miller et al. (1979) and Chigbo et al. (1982).

However, the metal root level ratios of dosed to control plant after 8 weeks are 200.8, 8.3, 7.8 and 5.1 for Cd, Cu, Zn and Pb respectively. This suggests that the uptake ability is in the order of Cd>Cu>Zn>Pb.

Interpretation of the variations in metal concentration the root to rhizome transfer mechanism, suggests that assuming negligible direct absorption, is most efficient followed by for Cu Pb with Cd and Zn and shows metal accumulation within preferential the root zone. Previous studies have demonstrated that up to 75% of added Zn tends to be deposited in Typha roots (Blake et al., 1987) and about 80% of the total Cd uptake was also found in Eichhornia crassipes (water hyacinth) roots by Blake et al. (1987).

Metal concentration values of <u>Typha</u> can be converted to tissue loadings using the dry weight biomass tissue ratios for <u>Typha</u> collected from Site 1 (root: rhizome: leaf = 1:18:14, see Chapter, 7); the metal distributions are as shown in Table 4.3. The metal loading data clearly indicate that the major metal bioaccumulation target area in <u>Typha</u> is the rhizome. It is interesting that for all metals, the root and leaf tissues show similar overall metal bioaccumulation loads. The leaf metal yields do have important implications for pollution control in terms of harvesting of the emergent foliage.

	Root	Rhizome	Leaf
Cd	662	1669	613
Cu	190	1188	329
Pb	242	976	532
Zn	689	1800	512

								مله
Table 4.3:	Metal	Loads	in	Typha	Tissue	for	Dosing	Experiment [^]

* values would be in mg if based on extraction from 1kg dry weight root tissue

4.2.2.2 Long term dosing experiments with Juncus effusus

Greenhouse-based, long term dosing experiments with Juncus were also carried out over an eight week period. During this test, peat and plants were dosed with (10 ppm) mixed Surface peat metal levels demonstrated metal solution. substantially enhanced metal levels over the eight week experimental period with increases observed rising from 0 to 314.0, 2 to 198, 11.2 to 203, and 19.2 to 323 mg/kg for Cd, Cu, Pb and Zn respectively (Fig.4.5). All metal levels in Juncus root tissues increased consistently if only at lower rates over the 8 weeks experimental period, but compared to the metal elevation patterns observed for the sediment, the root tissues patterns are more consistent. The metal elevation of plant leaf tissues showed a reduced affinity but nevertheless similar of patterns small consistent increases can be observed (Fig.4.5).

In comparison to the uptake of Cu and Pb, the roots show a higher uptake ability for Zn and Cd. After 8 weeks, metal concentrations increased by a factor of 2.8, 2.0, 1.6 and 1.5 for Cd, Zn, Cu and Pb respectively. This would correspond to Cd and Zn established equilibria noted in urban waters as being predominantly soluble bioavailable metal ions or as weak complexes (Revitt and Morrison, 1987).

The observed peat, <u>Juncus</u> root and leaf metal uptake data over the eight week time period of the dosing experiment have also been analysed by statistical regression methods. The resulting equations are shown in Fig.4.6. <u>Juncus</u> leaf Cu and Cd uptake patterns are exponential whereas Pb and Zn uptake patterns are linear. The root uptake trends for all metals are linear.





Dosing concentration 10 ppm Dosing time 3/10/1989 - 29/11/1989

Fig.4.5 Temporal increases in peat and <u>Juncus</u> tissue metal concentrations during 8 week dosing experiments



Fig.4.6 Regression equations and graphs of metal uptake by peat and Juncus tissues during 8 week dosing experiments

- 69 -

After the 8 weeks dosing experiment, the root to leaf ratios of <u>Juncus</u> plants decreased from 4.3 to 1.8, 3.2 to 1.8, 2.1 to 1.8, and 2.8 to 2.6 for Cu, Cd, Zn and Pb respectively. This suggests that the metals taken up by the roots are being transferred to the leaf efficiently and in the order of Cu>Cd>Zn>Pb. Previous studies have demonstrated the translocation of Cd from roots to above ground parts in water hyacinth (Blake et al., 1987).

4.2.2.3 Long term dosing experiment with peat

An additional 8 week dosing experiment was also carried out with peat to examine the metal uptake of the substrate material in the absence of plants. The results of this experiment are compared with the peat metal uptake data obtained from the long term dosing experiment with <u>Typha</u> <u>latifolia</u> (Fig. 4.7).

Surface peat concentrations are increased progressively for all metals (Fig.4.7) but are all lower than the surface peat metal levels recorded in the case of the dosing experiment conducted in the presence of <u>Typha</u>. The bottom peat concentrations of all metals also increased consistently with Zn and Cd levels equilibrating throughout the peat profile at end of the experiment. Basal peat Cu and Pb levels however remain lower than the surface levels.







Fig.4.7 Metal levels in surface and bottom peat in the presence or absence of <u>Typha</u> during 8 week dosing experiments

- 71 -

Comparison of the bottom peat metal levels showed that after 5 weeks, the metal levels of bottom peat without plant continued to increase, whereas peat with Typha levels decreased. Metal and associated water uptake by Typha roots is clearly resulting in the relatively lower observed bottom peat concentrations, and the decrease in levels after 5 weeks is possibly a consequence of an increase in ambient temperatures and also the evapotranspiration rate.

4.2.2.4 Comparison of dosing experiments with <u>Typha</u> <u>latifolia</u> and <u>Juncus effusus</u>

The results of the 8 week dosing experiments with <u>Typha</u> were compared with those obtained with Juncus (Fig. 4.8).

All metal levels in Typha root tissues show a constant uptake with Zn and Cd levels up to 689 and 662.5 mg/kg respectively. The metal root level ratios of 8 week dosed to controlled plant are 8.3, 5.1, 7.8 and 200.8 for Cu, Pb, Zn and Cd respectively (Table 4.2). In comparison to levels in the dosed Juncus root Typha, metal are relatively stable. The metal root level ratios of dosed to controlled Juncus at any dosing duration are lower than Typha. Although the experiments were not those for conducted simultaneously, the data indicate that Typha roots have a higher ability for metal uptake.





5-6

Root

3-4

Leaf

Duration of dosing experiment (weeks)

(c) Zinc

7-8

0

Control

0-2

3-4

Duration of dosing experiment (weeks)

Leaf 🖾 Root

(d) Cadmium

5-6

7-8

- 73 -

0

Control

Dosing concentration 10 ppm

0-2

Even though all metal levels in <u>Juncus</u> leaf tissues are higher than the levels recorded in <u>Typha</u> leaf tissues after 8 weeks, the metal ratios of dosed to controlled plant for Cu (except 7-8 weeks dosed/control), Pb and Cd are lower than the ratios of <u>Typha</u>. <u>Juncus</u> Zn level ratios of dosed to controlled plant are all higher than the ratios of <u>Typha</u>. This suggests that the translocation of Zn from root to leaf of <u>Juncus</u> is faster than for <u>Typha</u>, whereas Typha shows a higher affinity for Cd, Cu and Pb.

			root				
		Control	1-2	3-4	5 - 6	7-8 (weeks)	
Cu	Typha	1 (23.0)	1.9	3.6	4.8	8.3	
Pb	<u>Juncus</u> Typha	1 (39.0) 1 (47.2)	1.1 2.8	1.4 3.5	1.5 3.6	1.6 5.1	
Zn	Juncus Typha	1 (82.5) 1 (87.6)	1.2 1.5	1.4 1.5	1.5 3.9	1.5 7.8	
	Juncus	1 (52.0)	1.5	1.8	2.0	2.0	
Cđ	<u>Typha</u> Juncus	1 (3.3) 1 (30.0)	20.5	31.2	63.3 2.3	200.8	
	Juneus	1 (30.0)	1.0	200			
			leaf				
		Control	1-2	3-4	5 - 6	7-8 (weeks)	
Cu	Typha	1 (10.4)	1.8	2.6	2.9	2.3	
ЪĻ	Juncus	1 (9.0)	1.8	2.1 2.1	2.3	3.8 2.2	
PD	Juncus	1(1).2) 1(29.0)	1.4	1.5	1.6	1.7	
Zn	Typha	1 (23.2)	1.2	1.6	1.6	1.6	
6J	Juncus Typha	1(24./) 1(2.4)	1.8 2.1	2.3 6.5	2.3 15.2	18.3	
	Juncus	1 (9.5)	1.2	1.9	4.5	5.2	
							_

Table 4.4: Comparison of metal root and leaf ratios between controlled and dosed plant

4.3 Conclusions

The major findings of this section of the research can be summarised as follows:-

a) The short term greenhouse-based metal uptake experiments show that metal levels in dosed peat, <u>Typha</u> root, rhizome and leaf increase after 10 days and when the dosing water concentrations were increased, the concentration differences between peat, root, rhizome and leaf also increased in parallel. Dosed peat and <u>Typha</u> roots show a greater uptake ability for Cd and Cu than for Zn and Pb.

b) Long term greenhouse-based metal uptake experiments with Typha showed a greater enhancement of surface peat metal levels by over the eight week experimental period, whereas bottom peat metal levels remain relatively stable. The surface peat metal levels are up to 4.8 higher than basal metal levels. The surface sediment increases are most closely paralleled by metal uptake patterns in the Typha root, with Zn and Cd both exhibiting exponential increases with time and final values approaching 700 mg/kg. The uptake ability is in the order of Cd>Cu>Zn>Pb. The root to rhizome translocation is most efficient for Cu followed by Pb, with Cd and Zn showing a preferential metal accumulation within the root zone.

c) Long term dosing experiment with <u>Juncus</u> <u>effusus</u> also shows an enhancement of peat metal uptake, but metal elevation patterns of root tissues are more consistent over time. In comparison to Cu and Pb, roots translocation efficiency is in the order of Cu>Cd>Zn>Pb.

d) The long term dosing experiments with peat show that the surface peat concentrations of all metals increased constantly but are all lower than the surface peat metal levels recorded in the dosing experiments conducted in the presence of plants over a 1 week period. After 5 weeks, the metal levels of basal peat in the absence of the plant continue to increase, whereas the metal levels of bottom peat with plants decrease.

e) The comparison of dosing experiments with <u>Typha latifolia</u> and <u>Juncus effusus</u> which were not carried out simultaneously suggest that the former plants have a higher ability for metal uptake. The results also show that the metal translocation from root to leaf is more efficient in <u>Typha</u> for Cd, Cu and Pb whereas <u>Juncus</u> is more efficient for Zn translocation.

Chapter 5: Field Studies of Metal Uptake by Typha latifolia

5.1 Introduction

Man's activity has caused a drastic changes in metal fluxes within the biosphere (Grahn and Hakanson, 1987). In recent years, the fluxes of many trace metals from terrestrial and atmospheric sources to the aquatic substantially environment have increased (Forstner & Wittmann, 1981). When a trace metal is added to a natural water system, it partitions amongst various compartments. As a result of complex physical, chemical and biological processes, a major fraction of the trace metal introduced into the aquatic environment eventually becomes associated with the bottom sediment (Tessier and Campbell, 1987). Metals present in the water and sediment can then be taken up by aquatic macrophytes. Emergent plants obtain all their minerals from the sediment (Haslam, 1987). Studies have also shown seasonal variations to occur in metal uptake of the emergent species such as Phragmites sp (Larsen and Schierup, 1981; Mortimer, 1985).

The aim of this part of the study is to determine metal uptake under field conditions by sediment and <u>Typha</u> latifolia plants.

<u>Typha latifolia</u> plants and their associated sediment and surrounding waters within a rural pond (site 1) and two urbanized wetlands (sites 2 and 3) receiving surface discharges were analysed. Seasonal variations of Cd, Cu, Pb and Cd in sediment, water, and <u>Typha</u> root, rhizome and leaf tissues were determined. The metal levels in sediment were then compared with those recorded in plant root,

- 77 -

rhizome and leaf tissues.

5.2 Discussion of the Results

5.2.1 Heavy Metal Levels in Wetland Water Samples

Ranges and average metal concentrations monitored over a 14 month field study for the water phase are shown in Fig.5.1. The average Cu, Pb, Zn and Cd levels at the two urban sites (sites 2 and 3) are higher than those recorded at the rural site (site 1) with site 3 being the most polluted. The concentration differences of all metals between site 3 and site 1 are statistically significant (Table 5.1). The concentration differences of Cu and Cd between site 2 and site 1 are significant, whereas the differences of Pb and Zn are not. The ratios of mean metal levels for the sites (3:2:1) are: Cu, 4.8:2.9:1, Pb, 2.0:1.2:1, Zn, 2.0:1.3:1 and Cd, 2.2:1.4:1.

The water metal levels show a large seasonal variability at the two urban sites whereas the metal levels, with the exception of Zn, in water at site 1 are relatively stable (Table 5.2).



Sampling time: 12/10/1988 - 6/12/1989

Fig.5.1 Average and ranges of heavy metal concentration in wetland water samples

	Table 5.1: T values o	f water samples
	using Students	't' test
	t _{0.05,11} (site 1 and site 2)	t _{0.05,11} (site 1 and site 3)
Cu Pb Zn Cd	1.865 (>1.796) 0.865 (<1.796) 1.281 (<1.796) 3.324 (>1.796)	3.928 (>1.796) 1.914 (>1.796) 2.643 (>1.796) 1.904 (>1.796)

when n=11, P<0.05, $t_{0.05,11}>1.796$, the difference is significant

	Site 1	Site 2	Site 3
Cu	17.0 ± 21.3	39.4 ± 55.2	59.6 ± 46.0
РЪ	23.8 ± 22.3	27.4 ± 18.8	36.2 ± 15.7
Zn	63.0 ± 42.1	77.5 ± 40.4	136.6 ± 107.1
Cd	5.3 ± 4.5	6.0 ± 3.1	8.9 ± 4.2

Table 5.2: Average and standard deviation of metal levels in water samples (ug/1)

The overall temporal changes in water metal levels at the three sites and corresponding rainfall levels are shown in Fig.5.2 a-d. Copper, Pb and Cd levels at all sites show elevations in late spring/early summer which may be a consequence of storm events; this is most noticeable in the case of Pb and Cd. Copper levels at site 3 reach the highest value 116 ug/l in October following earlier lower summer peaks also associated with storm events. There is no clear trend in Zn levels which can be discerned at any of the sites although there is a tendency of a summer 'depression' in the observed values. The average Zn and Pb levels (107.3 and 39.8 ug/l) of the most polluted site (site 3) are similar to those recorded by Prahalad and Seenayya (1989) in a polluted lake.

In aquatic environments, water is the primary contaminated medium and an inherently high degree of variation in metal concentration naturally arises. Metals that do not remain soluble in the water phase become accumulated within the bottom sediment through sedimentation, thus the metal levels found in the sediment are affected by the concentration of metal levels in the ambient water.





Rainfall (mm)

D

S

A

Rainfall

Ν





5.2.2 Heavy Metal Levels in Wetland Sediments

Average metal concentrations together with the ranges monitored during the 14 month field study for sediment samples are shown in Fig.5.3. Copper, Pb, Zn and Cd levels exhibit the expected elevations at the two urbanised sites (sites 2 and 3) compared to the rural site (site 1). The differences in all metal levels between both urban sites and the rural site are statistically significant (Table 5.3). The ratios of mean metal levels for the sampling sites (3:2:1) are: Cu, 7.8:5.2:1, Pb, 10.8:7.0:1, Zn, 11.0:7.9:1 and Cd, 3.0:2.6:1.



Sampling time: 12/10/1988 - 6/12/1989

Fig.5.3 Average and ranges of heavy metal concentration in wetland sediment

<u>Table</u>	<u>5.3:</u>	<u>T</u>	values	of	sed	iment	samples
	usi	Ing	Stude	nts	't'	test	

	t _{0.05,11} (site 1 and site 2)	t _{0.05,11} (site 1 and site 3)
Cu	31.788 (>1.796)	14.037 (>1.796)
Pb	10.559 (>1.796)	10.947 (>1.796)
Zn	12.118 (>1.796)	18.171 (>1.796)
Cd	5.953 (>1.796)	4.667 (>1.796)

when n=11, P<0.05, $t_{0.05,11}>1.796$, the difference is significant

All sediment levels show a high variability at the two urban sites with the greatest variation being observed at site 3. Metal levels in the rural site sediment are very much more stable (Table 5.4). The average Cu and Zn levels of the two urban sites are similar to those recorded by Oliver (1973), Aulio (1980) and Prahalad and Seenayya (1989) in polluted rivers and lakes. These results are between McNaughton et al. (5000 mg/kg for Zn) (1974) and Taylor and Crowder's (3738 mg/kg for Cu, 343 mg/kg for Zn) (1983) results. The average Pb levels of two urban sites 841.2 and 545.4 mg/kg are higher than those recorded by Oliver (390 mg/kg, 1973), McNaughton et al. (435 mg/kg, 1974) and Prahalad and Seenayya (59.7 \pm 20.0 mg/kg, 1989). The sediment Cd levels at the two urban sites are much lower than those recorded by McNaughton et al. (73 mg/kg) (1974).

	Site 1	Site 2	Site 3
Cu	28.2 ± 7.9	119.9 ± 58.4	219.8 + 44.5
Pb	77.7 ± 15.6	496.2 ± 202.0	841.2 ± 228.8
Zn	70.7 ± 30.6	506.0 ± 197.1	778.9 ± 139.6
Cd	4.2 ± 3.3	9.7 ± 4.8	12.5 ± 7.8

Table 5.4: Average and standard deviation of metal levels in sediment samples (mg/kg)

There is field evidence for contaminated sediment flushing occurring to the receiving basin (site 3) during late summer/early autumn following storm events, especially in the case of Cu, Pb, and Cd (Fig.5.4 a-d). This follows the early summer peak levels of Cu, Pb and Cd observed in the water phase. These events appear to have entrained and incorporated toxic materials within the mud substrate of the receiving basin. Zn sediment levels stay relatively stable during this period at all sites, corresponding to the trends noted in the water levels.

Sediments play a crucial role in both ambient water quality and in plant metal uptake. They act as a sink to remove metal pollutants from the water column and may release metal contaminants to surface waters. The amount of toxic heavy metals present in contaminated sediment is considered to be the amount potentially available to the plants. Emergent plants obtain all their minerals from the sediment (Haslam, 1987), therefore a proper assessment of metal levels in the sediment is essential when considering potential metal uptake by aquatic plants.





85 **1**

5.2.3 Heavy Metal Levels in Typha Root Tissues

The highest <u>Typha</u> metal levels are recorded in the root tissue. Average metal concentrations together with ranges monitored during the 14 month field study for <u>Typha</u> root tissues samples are shown in Fig.5.5. Corresponding to the sediment metal levels, Cu, Pb, Zn and Cd levels exhibit the expected elevations at the two urbanised sites (sites 2 and 3) compared to the rural site (site 1). The differences of all metal levels between urban sites and rural site are statistically significant (Table 5.5). The metal ratios for the sampling sites (3:2:1) are: Cu, 1.9:1.4:1, Pb, 2.0:2.0:1, Zn, 1.9:1.9:1 and Cd, 1.8:1.5:1.



Sampling time: 12/10/1988 - 6/12/1989

Fig 5.5 Average and ranges of heavy metal concentration in <u>Typha</u> root tissues

	t _{0.05,11} (site 1 and site 2)	t _{0.05,11} (site 1 and site 3)
Cu	4.386 (>1.796)	4.865 (>1.796)
Pb	3.382 (>1.796)	3.685 (>1.796)
Zn	4.795 (>1.796)	10.371 (>1.796)
Cd	4.978 (>1.796)	6.377 (>1.796)

Table 5.5: T values of Typha root samples using Students 't' test

when n=11, P<0.05, $t_{0.05,11}$ >1.796, the difference is significant

The Cu, Pb, Zn and Cd levels in root tissues show considerable temporal variability (Table 5.6), with peak levels of Cu occurring during the growth period which may be associated with an expected increase in demand for plant micronutrients (Fig.5.6 a). The maximum Pb and Cd root levels occur during August to October following bioaccumulation (Fig.5.6 b,d), As in the case of sediment and water, Zn concentrations in the plant root tissues at all three sites do not demonstrate any distinct seasonal variations (Fig.5.6 c) as do the other metals. However, Zn corresponding with concentrations fall in May translocation to rhizome and leaf during growth. The average root Pb levels at the two urban sites are similar to those obtained from Potamogeton robbinsii root by Reimer (1989).

	Site 1	Site 2	Site 3
 Cu	23.0 + 14.7	32.7 + 17.7	39.7 ± 17.7
Pb	47.3 ± 23.5	94.5 ± 59.4	85.6 ± 56.1
Zn	87.5 ± 17.6	160.1 ± 52.6	153.5 ± 58.2
Cd	3.3 ± 1.5	5.0 ± 2.2	5.4 ± 2.9

Table 5.6: Average and standard deviation of metal levels in Typha root samples (mg/kg)

Root Cu and Zn levels at the two urban sites are within the range of the results (13 - 265 mg/kg for Cu and 24 -572 mg/kg for Zn) of a study of metal accumulation of Typha latifolia near smelters (Taylor, 1983).

Emergent plants like <u>Typha latifolia</u> absorb relatively few nutrients or metals though their thick rhizomes or aerial parts (Haslam, 1987). Root functions therefore play a crucial role in metal uptake for emergent plant species.



~



- 68 -

5.2.4 Heavy Metal Levels in Typha Rhizome Tissues

metal Average concentrations together with ranges monitored during the 14 month field study of Typha rhizome samples are shown in Fig. 5.7. The average metal levels in rhizome tissues are less than the metal levels in root but greater than those in leaf tissue. Copper, Pb, Zn and Cd levels at the two urbanised sites (site 2 and 3) are consistently higher than those at the rural site (site 1). The differences of all metal levels between urban sites and the rural site are statistically significant (Table 5.7). The ratios of the average metal levels in Typha rhizome tissues for the sites (3:2:1) are: Cu, 1.8:1.5:1, Pb, 1.4:1.2:1, Zn, 1.8:1.4:1 and Cd, 1.5:1.3:1.



Sampling time 12/10/1988 - 6/12/1989

Fig.5.7 Average and ranges of heavy metal concentration in plant rhizome tissues

	t _{0.05,11} (site 1 and site 2)	t _{0.05,11} (site 1 and site 3)
Cu	3.326 (>1.796)	3.180 (>1.796)
Pb	2.852 (>1.796)	3.427 (>1.796)
Zn	2.805 (>1.796)	4.214 (>1.796)
Cd	2.401 (>1.796)	2.695 (>1.796)

Table 5.7: T values of Typha rhizome samples using Students 't' test

when n=11, P<0.05, $t_{0.05,11}$ >1.796, the difference is significant

All metals show temporal variability (Table 5.8) in <u>Typha</u> rhizome tissues with peak levels of Cu and Zn occurring during the growth period (Zn in June; Cu in July) and thus with an increase in demand for these associated plant micronutrients (Fig.5.8 a,c). There is general bioaccumulation of Cd and Pb in the rhizome during July to October (Fig.5.8 b,d). Rhizome Cu and Zn levels at the two urban sites are very similar to those obtained by Taylor and Crowder (1983) in <u>Typha latifolia</u> and Aulio (1980) in <u>Nuphar lutea</u> near a metal processing plant.



- 92 -



	Site 1	Site 2	Site 3
Cu	12.6 ± 12.9	19.0 ± 18.8	22.7 + 21.2
Pb	21.5 ± 14.7	26.6 ± 13.7	31.6 ± 18.2
Zn	25.4 ± 12.1	34.3 ± 12.5	46.1 ± 21.2
Cd	2.6 ± 1.6	3.3 ± 1.7	3.9 ± 2.0

Table 5.8: Average and standard deviation of metal levels in Typha rhizome samples (mg/kg)

Plant rhizomes also participate in metal uptake from sediment although emergent species cannot significantly absorb metals directly through their thick rhizomes (Haslam, 1987). However, rhizomes play an important part in metal storage (see Chapter 6).

5.2.5 Heavy Metal Levels in Typha Leaf Tissues

metal concentrations together with ranges Average monitored during the 14 month field study for the leaf samples are shown in Fig.5.9. Although the metal leaf levels at the two urbanised sites (site 2 and 3) are higher than those of the rural site (site 1), the site 3 to site 1 mean concentration ratios decrease in comparison to root and rhizome. The ratio of metal levels in Typha sites (3:2:1) are: Cu, 1.5:1.1:1, Pb, leaf for the 1.4:1.4:1. Cd, The and 1.6:1.1:1 Zn, 1.1:1.2:1, differences in all metal levels between site 1 and site 3 are statistically significant. The differences in Pb and Cd levels between site 1 and site 2 are significant whereas Cu and Zn levels are not (Table 5.9).



Sampling time: 12/10/1988 - 6/12/1989

Fig.5.9 Average and ranges of heavy metal concentration in <u>Typha</u> leaf tissues

	Table 5.9: T values of T	ypha leaf samples
	using Students	't' test
·		
	t _{0.05,11} (site 1 and site 2)	t _{0.05,11} (site 1 and site 3)
Cu Pb Zn Cd	1.591 (<1.796) 2.248 (>1.796) 0.685 (<1.796) 2.640 (>1.796)	3.993 (>1.796) 2.112 (>1.796) 2.654 (>1.796) 2.888 (>1.796)

when n=11, P<0.05, $t_{0.05,11}$ >1.796, the difference is significant
All metals show temporal variability (Table 5.10) in Typha leaf tissues, with peak levels of Cu and Zn occurring during the growth period when both root and rhizome demand for these associated plant micronutrients is highest (Fig. 5.10 a,c). There is a progressive bioaccumulation of Cd in leaf tissue during May to October with peak levels being achieved in October (Fig. 5.10 d). Maximum leaf levels of Pb at all sites occur during August to October possibly due to translocation from the root tissues being by atmospheric deposition augmented as well as by increases in the permeability of the leaf during decomposition (Fig.5.10 b). This latter mechanism has been by Larsen and Schierup (1981) suggested to work effectively for Phragmites which also showed temporal variability of Zn, Cu and Pb in leaf tissue.

	Site 1	Site 2	Site 3
Cu	10.4 ± 13.7	11.5 ± 15.1	15.1 ± 14.8
Pb 7n	17.2 ± 15.0	20.1 ± 15.9	18.8 ± 15.3
Cd	2.4 ± 1.6	3.4 ± 2.0	3.4 ± 1.9

Table 5.10: Average and standard deviation of metal levels in Typha leaf samples (mg/kg)



Fig.5.10 Temporal variations of heavy metal levels in Typha leaf tissues

- 96 -

The maximum leaf Pb levels (site 3) of 46.5 - 49.5 mg/kg are similar to those recorded by Reimer (1989) in a study which also showed considerable inter specific variation in macrophyte leaf Pb uptake. The average Cu levels of 15.1 and 11.5 mg/kg found at the two urban sites are similar to those recorded by Taylor and Crowder (1983). Zn levels 37.1 and 26.1 mg/kg at the two urban sites are higher than Taylor's results (18 \pm 1 mg/kg) for a field study of metal uptake by Typha latifolia located near smelters. The average Cu level 10.4 mg/kg of site 1 are in the range of previous studies (of uncontaminated sites) on Typha latifolia and Typha angustifolia by Boyd (34 mg/kg, 1970 a), Mudroch and Capobianco (1.3 mg/kg, 1978), Mudroch (3-4 mg/kg, 1980), Reimer and Toth (6.3 mg/kg and 5.0 mg/kg, 1968), Varenko and Chuiko (24.3 1971) mg/kg, and Guilizzoni (4.8 mg/kg, 1975).

Emergent species absorb little minerals through their aerial (leaf) parts with the main metal load found in plant leaf tissues arising from root uptake. The leaf metal yields have important implications for pollution control in terms of harvesting of the above-water foliage and this will be discussed later.

5.2.6 Comparison of metal levels in water, sediment and plant tissues

The concentrations of all metals in sediment and <u>Typha</u> are considerably higher than in the surrounding water. The ratios of sediment and <u>Typha</u> metal levels to surrounding water levels at site 3 are given in Table 5.11. The metal level ratios of <u>Typha</u> tissues to water range from 283 to 2389 whereas metal level ratios of sediment to water are between 1393 and 21136. These results are consistent with those suggested by Merchyulenene and Nyanishkene (1976) and Leland and McNurney (1984).

	Water (ug/l)	Root	Rhizome	Leaf	Sediment
Cd	1 (8.9)	674	438	382	1393
Cu	1 (53.4)	805	425	283	4137
Pb	1 (39.8)	2389	794	472	21136
Zn	1 (107.3)	1589	430	346	7259

Table 5.11: Metal level ratios of sediment and plant to water (site 3)

The average metal concentrations monitored during the 14 month field study of sediment, root, rhizome and leaf samples at the three sites are shown in Fig.5.11.

There is a progressive decrease in Cu, Pb, Zn and Cd concentrations from sediment to <u>Typha</u> root, to rhizome and to leaf at all sites (except Zn at site 1). This is consistent with Taylor and Crowder's studies of Cu and Zn levels in <u>Typha latifolia</u> (except for Zn levels in leaf, 1983). The ratios of sediment to root to rhizome to leaf concentrations are given in Table 5.12. Welsh and Denny (1980) and Heisey and Damman (1982) have reported Pb root to leaf ratios of between 3.8 - 5.8 and 3 respectively for submerged macrophytes. In the present study root to leaf ratios are 2.7, 4.7 and 5.1 for site 1, 2 and 3 respectively.



S:sediment, R:root, Rh:rhizome, L:leaf Sampling time 12/10/1988 - 6/12/1989

Fig.5.11 Heavy metal levels in sediment and Typha tissues

- 99 -

		sediment	:	root	:	rhizome	: leaf
Site 3	Cu Pb Zn Cd	14.6 44.8 21.0 3.6		2.8 5.1 4.6 1.8		1.5 1.7 1.2 1.1	(mg/kg) 1 (15.1) 1 (18.8) 1 (37.1) 1 (3.4)
Site 2	Cu Pb Zn Cd	12.8 27.1 21.3 3.5		2.9 4.7 6.4 1.7		1.7 1.3 1.3 1.1	$ \begin{array}{c} 1 (11.5) \\ 1 (20.1) \\ 1 (26.1) \\ 1 (3.4) \end{array} $
Site 1	Cu Pb Zn Cd	2.7 4.5 3.0 1.7		2.2 2.7 3.8 1.4		1.2 1.3 1.1 1.1	$ \begin{array}{c} 1 & (10.4) \\ 1 & (17.2) \\ 1 & (23.2) \\ 1 & (2.4) \end{array} $

Table 5.12: Sediment to leaf metal average concentration ratio

Although root to leaf mean concentration ratios increase for Cu, Pb, Zn and Cd at the urban sites, the ratios of rhizome to leaf concentrations are relatively constant for all metals at the three sites which range between 1.1:1 for Cd to 1.7:1 for Pb and Cu. This suggests that the ultimate tissue distribution is probably constitutionally determined and is independent of root metal uptake rates. Taylor and Crowder (1983) have shown that Zn and Cu concentrations of <u>Typha</u> leaves were not correlated with sediment concentrations.

Sediment to leaf mean Pb concentration ratios increase in the order of 4.5:1, 27:1 and 44.8:1 at sites 1, 2 and 3 respectively, indicating the affinity of Pb for the sediment particulate phase. However, a similar trend is observed for Zn, which is known to occur predominantly in the more bioavailable dissolved phase. The sediment to leaf mean metal concentration ratios are in the order of Pb>Zn>Cu>Cd at all sites. This implies that the uptake ability of <u>Typha</u> plants is in the reverse order of Cd>Cu>Zn>Pb.

5.3 Conclusions

The major findings of this section of the research can be summarised as follows:-

a) An increase in sediment and <u>Typha</u> tissue levels is observed from the rural to urban sites with site 3 to site 1 sediment average concentration ratios ranging from 3:1 for Cd to 11:1 for Zn. The range of site 3 to site 1 root average concentration ratios varies from 1.8:1 for Cd to 2:1 for Pb.

b) The statistical analysis shows that the water concentration differences of all metals between site 3 and site 1 are significant. The concentration differences of Cu and Cd between site 2 and site 1 are significant whereas the differences of Pb and Zn are not. The metal concentration differences in sediment, Typha root and rhizome between urban sites and the rural site are also significant as are the leaf concentration differences of al1 metal levels between site 1 and site 3. The differences of Pb and Cd levels between site 1 and site 2 are significant whereas Cu and Zn levels are not.

c) Field samples from all sites show a progressive decrease in metal levels from sediment to root to rhizome

-101-

to leaf, with sediment to leaf ratios increasing in the case of Pb from 4.5:1 to 44.8:1 from site 1 to 3. Although root to leaf mean concentration ratios increase for Cu, Pb, Zn and Cd at the urban site, the ratio of rhizome to leaf concentration is relatively constant for all metals at all sites. This suggests that the ultimate tissue distribution is probably metabolically determined and is independent of root metal uptake rates. The sediment to leaf mean metal concentration ratios are in the order of Pb>Zn>Cu>Cd at all sites. This implies an uptake ability of Typha plants in the reverse order of Cd>Cu>Zn>Pb.

d) All metals show temporal variability in Typha tissues, with peak levels of Cu and Zn in root, rhizome and leaf coinciding with the growth period and thus with these associated plant increase demand for in an is bioaccumulation of Cd and РЪ There micronutrients. during August and October in root, rhizome and leaf tissues. The temporal variability in accumulation rates, leaf tissue, has important management especially for implications in terms of heavy metal removal through by harvesting the standing leaf crop.

<u>Chapter 6:</u> The Comparison of Field Studies and Greenhouse Dosing Experiments and their Management Implications

6.1 Introduction

Emergent aquatic macrophytes can take up heavy metals from sediments via root systems and translocate them to the rhizome and aerial parts. However, the greater part of heavy metals taken up by the plants is retained in the roots and rhizome with smaller amounts being translocated to the above - ground sections of the plant (Larsen and Schierup, 1981). Heavy metals stored in the above - ground part can be removed by cropping the leaves. Black et al. (1979, 1984) reported that cutting and harvesting Typha latifolia leaves can remove 25 to 42 % of the Zn accumulated in the plants and prevents the leaching of the metal from associated decaying organic matter.

The aim of this study was to compare metal loads found in <u>Typha latifolia</u> resulting from both greenhouse-based dosing experiments and from field studies. Heavy metal distributions found in <u>Typha</u> have already been analysed and the management implications of heavy metal removal by cropping plant leaves are discussed further in this section.

6.2 The comparison of field studies and greenhouse experiments

6.2.1 The comparison of sediment metal levels

The average metal levels in the sediments at three field sites and metal levels in the dosed peat from greenhouse

experiment after 8 weeks have been compared with the unpolluted and toxic levels found in soil which were given in Table 2.2 (Romero et al., 1989 and Baker and Brooks, 1989). An evaluation of toxic metal levels in sediment can be derived from inspection of Table 6.1. All metal levels, with the possible exception of Cd, found at site 1 are well below toxic levels with Cu and Zn levels being very close to the unpolluted metal levels of background soils. Sediment Cu, Pb, Zn and Cd levels at both urban sites are very much higher than the generally quoted toxic levels with the highest ratio (metal levels of urbanised sites investigated to general toxic levels in soil) being for Pb (5.5 and 8.4 at site 2 and 3, respectively). This reflects the generally more contaminated and toxic nature often associated with urban discharges (Ellis, 1986). The relative toxicities of the metals are in the order of Pb>Cd>Zn>Cu for site 2, and Pb>Cd>Cu>Zn for site 3.

	Soil (mg/	/kg)		Sedimer	nt investigated	(mg/kg)
	Unpolluted' levels	[*] Toxic [*] levels	Site	1 Site 2	Site 3	Dosed Peat (Surface)
Cu Pb Zn Cd	25 25 65 1	100 100 300 5	28 78 71 4	Ratio ⁺ 147(1.5) 545(5.5) 557(1.8) 11(2.2)	Ratio ⁺ 220(2.2) 841(8.4) 779(1.6) 12(2.4)	Ratio ⁺ 187(1.9) 168(1.7) 295(1.0) 286(57.2)

	Table	6.1:	The	toxicity	of	metal	levels	in	sedimer
--	-------	------	-----	----------	----	-------	--------	----	---------

* from Table 2.2

+ ratios of site 2 : toxic, site 3 : toxic and dosed : toxic

Copper, Pb and Cd levels in dosed peat after the 8 weeks dosing experiment are all higher than threshold toxic levels especially for Cd. Pb levels are much lower than the levels found at the two urban sites. The Cu is similar to the levels found in urban sites whereas Cd level is much higher than the urban sites. Zn levels in dosed peat are similar to the toxic level and fall between the levels of the rural and urban sites. The relative toxicity of the metals in dosed peat is in the order of Cd>Cu>Pb>Zn.

The maximum metal loads found in the sediments of the greenhouse experiments and field studies are given in Table 6.2. Copper, Pb and Zn levels for the two urban sites sediments are very much higher than the peat metal levels obtained following the 8 week dosing experiments (Cu levels at site 2 are similar to the results obtained for the dosing experiments). Cd levels however are much lower than the levels obtained for the dosed peat. This suggests that the input rates of metal loads received at the urban sites are high and exceed the potential rate of uptake and removal by <u>Typha</u> and therefore the metals accumulate in the sediment phase.

	Dosed (surface)	Site 3	Site 2	Site 1
Cu	187	304	178	42
Pb	168	1199	784	97
Zn	295	978	764	123
Cd	286	30	15	13

Table 6.2: Maximum Metal Loads in Sediment for Dosing Experiments and Field Studies

* values would be in mg if based on extraction from 1kg dry weight ie mg/kg equivalents

6.2.2 The comparison of Typha tissue metal levels

The average metal levels in plant tissues at three field sites and from greenhouse dosing experiments have been compared with the general unpolluted and toxic levels found in plants which were quoted in Table 2.2 (Table 6.3). Only Cu and Cd levels at site 3 reach the general toxic levels whilst all others fall below the indicated toxic levels. This is probably due to the avoidance or restriction of metal uptake in tolerant races of plant (Baker and Walker, 1990). The rhizome and leaf metal levels for field sites are all below the general toxic levels. Metal levels in dosed plant root are however much higher than the general toxic levels in the plant; Cu and Cd levels in Typha rhizome and Cd levels in the leaf also This demonstrates the high exceeded the toxic level. tolerance of metal toxicity for these species.

Plant (mg/kg)	Cu	Pb	Zn	Cd
Unpolluted*	12	10	60	
Toxic levels*	40	150	400	5
Root Site 1 Site 2 Site 3 Dosed	23.0 32.7 39.7 190.0	47.3 94.9 85.6 242.0	87.5 160.1 153.5 689.0	3.3 5.0 5.4 662.5
Rhizome Site 1 Site 2 Site 3 Dosed	12.6 19.0 22.7 66.0	21.5 26.6 31.6 54.2	25.4 34.3 46.1 99.5	2.6 3.3 3.9 92.7
Leaf Site 1 Site 2 Site 3 Dosed	10.4 11.5 15.1 23.5	17.2 20.1 18.8 38.0	23.2 26.1 37.0 36.6	2.4 3.4 3.4 43.8

Table 6.3: The toxicity of metal levels in plant

•

* from Romero et al., 1989, Baker and Brooks, 1989

The maximum metal loads in <u>Typha</u> root, rhizome and leaf for the greenhouse experiments and field studies are given in Fig. 6.1. The maximum root Pb levels at the two urban sites are similar to the root Pb levels obtained after the 8 weeks dosing experiment, whereas Cd, Cu and Zn levels are much lower than those of the dosed root. The maximum Cu, Pb and Zn levels in <u>Typha</u> rhizome and leaf levels of the two urban sites are similar to those obtained from the greenhouse dosing experiments, whereas the Cd levels are much lower.



D:Dosed, 3:Site 3, 2:Site 2, 1:Site 1

Fig.6.1 The comparison of <u>Typha latifolia</u> tissue metal levels in dosing experiments and field studies

The comparison of maximum <u>Typha</u> tissue metal levels for the field studies with the long term dosing experiments suggests that the limit of uptake for Cu, Pb and Zn is probably achieved at the urban sites (except Cu and Zn levels in root tissue) but has not yet been achieved in the case of Cd.

6.2.3 Metal uptake efficiency of Typha latifolia

Both short term and long term greenhouse-based metal uptake experiments with <u>Typha latifolia</u> show that the efficiency of <u>Typha</u> root metal uptake is in the order of Cd>Cu>Zn>Pb. The field studies of <u>Typha</u> metal uptake show the same results. These results are in agreement with those obtained by Harrison and Chirgawi (1989). In his study of soil and air as contributors of trace metal to terrestrial plants, it was suggested that the efficiency of uptake of metals from soil was generally high for Zn and Cd but low for Pb.

6.3 Management implications for heavy metal removal

6.3.1 Biomass measurement

The biomass of root, rhizome and leaf of <u>Typha</u> from site 1 during October. 1989 were measured and the results are shown in Table 6.4 below. The ratios of tissue biomass show that <u>Typha</u> root tissue biomass is relatively low in comparison to rhizome and leaf; hence its negligible role as a metal storage tissue. The highest biomass was found in <u>Typha</u> rhizome tissue. The rhizomes play an important role in metal storage (see section 6.3.2) with 42.4% of <u>Typha</u> plant biomass coming from leaves. This has important implications for metal removal by cropping (see section 6.3.3). The calculated leaf biomass (8381 kg/ha) and plant total biomass (19,761 kg/ha) are similar to those obtained by Blake et al. (9150 and 16,169 kg/ha for leaf and plant total biomass respectively), for their study of <u>Typha</u> <u>latifolia</u> on sediment contaminated with Zn (1987).

Biomass	kg/ha	Ratio	% of total	
 Root	604	1	3	
Rhizome	10.776	18	54.5	
Leaf	<i><u> </u> </i>	14	42.5	
Total	19,761	33	100	

Table 6.4: Dry Weight Tissue Biomass (October, 1989, Site 1)

The <u>Typha</u> leaf biomass (site 1) during the growing season was also measured and the results are shown in Table 6.5. The leaf tissue biomass increases from May to October with the highest leaf tissue biomass being found in July.

Table	6.5:	Dry Weig	ht Lea	<u>Tissue</u>	Biomass	(1989,	Site	<u>1)</u>

	May	June	July	October	
Leaf (kg/ha)	2400	7433	8740	8381	

6.3.2 Metal distribution in plant tissues

When metal levels are converted to tissue loadings using the dry weight biomass tissue ratios for <u>Typha</u> collected from the field sites (root:rhizome:leaf=1:18:14, Table 6.4), the resultant metal distributions are as shown in Fig.6.2.

The metal loading data clearly indicate that the major metal bioaccumulation target area in Typha is the rhizome with 54-61 % metal load being stored in this area. Approximately one-third of the metal load is stored in Typha leaf whereas only 4-12 % is stored in Typha roots. These results show that the major burden of heavy metal taken up by Typha is retained in roots and rhizomes (60, 67, 68 and 66 % for Cd, Cu, Pb and Zn respectively). This is in broad agreement with Blake et al.'s (1987) study which showed 58-75 % Zn of accumulated by Typha being stored in roots and rhizomes. However, Larsen and Schierup's study (1981) with Phragmites showed a much higher metal storage in roots and rhizomes (77-97 %). Typha accumulates some 32-40 % of metals in leaf tissue (Fig.6.2) and these metals could potentially be released by the decomposition processes (Black, 1979, 1984). It might therefore be necessary to cut and harvest the plant, as this technique would allow the export of 30-40 % of the metals accumulated in Typha latifolia and prevent the leaching from decaying organic matter of the metal back into the water phases.

- 111 -





Fig.6.2 Metal load distributions in Typha tissue (%)

6.3.3 Heavy metal removal

The partial removal of heavy metal pollutants in macrophytes can be achieved by harvesting of the foliage where the accumulated metals are stored. The variation the optimal cropping months for each metal, assuming one crop per year, is given in Table 6.6.

These results have some important management implications. annual crop of emergent An leaf biomass, preferably undertaken between June to October, would provide an efficient method of heavy metal removal from recently constructed wetlands. In mature wetlands however. preferential metal accumulation in the sediment store will inevitably reduce the value of metal removed by leaf harvesting. This is due to the unavoidable disturbance of the benthal substrates and consequent release of metals which will subsequently effect the potential advantages that might accrue from metal by cropping. removal Furthermore, any disruption of the submerged leaf tissue matrix would interfere with the absorption of metals from the water phase and their transfer to the rhizome, which Mortimer (1985) considers to be an important pathway.

	Cd	Cu	РЪ	Zn	
Optimal Cropping Month	October	July	October	June	
Site 1 Site 2 Site 3	25.1 58.7 54.5	440.5 494.7 524.4	389.7 406.5 414.9	275.0 308.5 680.0	

Table 6.6: Maximum Dry Weight Leaf Metal Loads (g/ha)

6.4 Conclusions

The major findings of this section of the research can be summarised as follows:-

a) The field study of sediment and plant samples indicates that all metal sediment levels at the urbanised sites are higher than the generally recorded toxic level. The toxic levels are found in the order of Pb>Cd>Zn>Cu for site 2, and Pb>Cd>Cu>Zn for site 3. These results indicate that <u>Typha latifolia</u> is a metal tolerant species and can thrive on metal contaminated sediment in urban wetlands. All sediment metal levels at the background control site 1 are below general background toxic levels with Cu and Zn levels being very close to the unpolluted metal levels found in soil.

b) Copper and Cd root levels at site 3, as well as Cd root levels at site 2 reached metal levels in <u>Typha</u> tissues which are regarded as potentially toxic to plants. Metal levels in dosed plant root are all much higher than the general toxic levels; Cu and Cd levels in <u>Typha</u> rhizome and Cd level in leaf also exceed the toxic level. This indicates that <u>Typha latifolia</u> cannot only survive in metal contaminated sediment but also can efficiently and rapidly uptake metals especially Cu and Cd to high levels.

c) The comparison of greenhouse experiments and field studies shows that under field conditions, heavy metals can accumulate in sediment to high levels. This result demonstrates the important role of the sediment store as a metal sink in polluted receiving waters. d) This study also shows that the limit of <u>Typha latifolia</u> metal uptake for Cu, Pb and Zn is probably achieved at urban sites over fairly short time scales, but it is rather more difficult to achieve saturation in terms of Cd uptake. The metal uptake efficiency of <u>Typha</u> roots for both greenhouse experiments and field studies show the order to be Cd>Cu>Zn>Pb.

e) The biomass measurement and plant metal distribution analysis show that <u>Typha</u> rhizome is the prime target area for metal storage (54 - 61 %). Plant leaves can store up to 32 - 40 % of metals accumulated in the plant tissue. These results can have important management implications.

f) Heavy metal removal can be achieved by cropping plant leaves. An annual crop of <u>Typha</u> leaf biomass, preferably undertaken in June to October can remove up to Cu 524, Pb 415, Zn 680 and Cd 59 g/ha. This could provide a useful method of heavy metal removal from recently constructed wetlands.

Chapter 7: Conclusions

7.1 Summary of major findings

a) The research shows an efficient metal uptake (Cu 25.1, Pb 20.3, Zn 32.5 and Cd 31.5 mg/kg/wk) by substratum(peat) in greenhouse-based phytoassays which is confirmed by the very high metal levels determined in urban field sediment studies. Levels found in urban sediment are $10^3 - 10^4$ times higher than the metal levels encountered in the water column and 1.5 - 8.4 times higher than the reported sediment toxic levels. This indicates that sediment plays a very important role as both a metal sink and reservoir for leaf decomposition as well as for contaminants contained in urban surface runoff discharges.

b) The research demonstrates effective heavy metal Cu, Pb, Zn and Cd uptake by <u>Typha latifolia</u>. The uptake efficiency of <u>Typha</u> roots for both greenhouse experiments and field studies were found to be in the order of Cd>Cu>Zn>Pb. The study also indicates that the limit of <u>Typha latifolia</u> metal uptake as determined by dosing experiment for Cu, Pb and Zn is probably achieved at the urban sites (except Cu and Zn levels in root tissue), but is not yet reached for Cd.

c) Both greenhouse experiments and field studies show that metal levels in <u>Typha latifolia</u> can reach levels which exceed the suggested toxic metal levels of plant tissues. The tolerance of metals is in the order of Cd>Cu>Pb>Zn. This indicates that <u>Typha latifolia</u> plants not only can survive in a highly metal contaminated sediment matrix but can also accumulate heavy metals especially Cd and Cu to high levels in the plant tissues. These results

- 116 -

demonstrate the high tolerance these species have for the metal toxic levels normally encountered in the environmental ecosystem.

d) All metals show marked temporal variability in <u>Typha</u> tissues, with peak levels of Cu and Zn found during the growth period when there is an increase in demand for these associated plant micronutrients. There is bioaccumulation of Cd and Pb during August and October in plant tissues. This temporal variability in metal levels especially for leaf components has important management implications by means of cropping plant leaves for heavy metal removal from urban wetlands.

e) The comparison of dosing experiments of <u>Typha latifolia</u> and <u>Juncus effusus</u> shows that <u>Typha</u> roots have a higher potential for metal uptake. The results also show that the metal translocation from root to leaf is more efficient in <u>Typha</u> for Cd, Cu and Pb, whereas <u>Juncus</u> is more efficient for Zn translocation.

f) The biomass measurements and plant metal distribution analysis show that <u>Typha</u> rhizome is the major target area for metal storage. <u>Typha</u> rhizomes can store 54 - 61 % of the metals accumulated in the plant tissues; plant leaves store some 32 - 40 % metals.

g) Heavy metal removal from urban wetlands can be achieved by cropping plant leaves. An annual crop of <u>Typha</u> leaf biomass, preferably undertaken in June - October can remove up to Cu 525, Pb 415, Zn 680 and Cd 59 g/ha. This can provide a useful method of heavy metal removal especially in recently constructed artificial wetlands. h) The results of this study provide useful information and methods for water quality improvement of urban runoff, which are especially valuable in terms of bioengineering design of urban flood storage ponds.

7.2 Suggestions for further work

a) The heavy metal concentrations presented in this study are of total metal concentrations. The sediment - metal speciation and their relationship with plant - metal uptake have not been studied. The study of such sediment metal speciation would provide further useful information about bioavailable metals present in sediment and reveal more detailed relationships on plant - metal uptake.

b) Chapter 4 has outlined the results of greenhouse dosing experiments carried out with static tanks. Further studies should be taken to examine heavy metal uptake by sediment and plant under flow-through conditions in both the laboratory and field. This would model heavy metal uptake in urban aquatic ecosystems and their mechanisms under more realistic and mobile conditions.

c) The studies described in chapter 5 have indicated the metal uptake ability of <u>Typha</u>. In order to understand the heavy metal cycle in an aquatic environment, it is very important to know the plant decomposition process and heavy metal release. Further work should be under taken to study the release mechanisms and controls of heavy metals from decomposing plant materials under both laboratory and field conditions.

d) The studies described in chapter 6 have indicated that the limit of <u>Typha latifolia</u> metal uptake examined by

dosing experiments for Cu, Pb and Zn is probably achieved at urban sites, but is not yet reached for Cd. Further work should be taken to study the factors which effect the plant metal uptake under both greenhouse dosing experiment and field conditions.

e) In order to examine the performance of macrophytes for metal uptake in urban wetlands and their role for water quality improvement in urban areas, controlled field site studies should be undertaken taken to determine the role of planted macrophytes species in both metal uptake and removal in artificially constructed urban wetlands. These studies could provide useful criteria for bioengineering design and urban wetlands management.

Acknowledgements

Many people have given me useful assistance throughout the research period. I would like to acknowledge all their contributions and especially the following:

Mr R.B.E. Shutes, my director of studies, Prof. J.B. Ellis and Dr D M.R. Revitt, supervisors, who provided much valuable advice and guidance throughout the project.

Many staff within the Urban Pollution Research Centre, and especially technican Desmond Bannon who gave me useful help with the practical work, and also my fellow research students.

The research project was funded by the Chinese National Education Committee, The Henry Lester Trust, The Great Britain-China Educational Trust, The Universities' China Committee in London, Middlesex Polytechnic Urban Pollution Research Centre and Liaoning Environmental Protection Institute.

I would also like to thank all the members of my family, especially my husband for their constant encouragement, understanding and support.

This thesis is dedicated to my daughter Wang Xiaopei who is living with grandparents in China in the hope she will understand why she has missed her mother's love in the last two years and also in the hope she may grow up in a world which is environmentally friendly and full of freedom and happiness.

REFERENCES

Abdalla, A.L., Ambrosano, E.J., Vitti, D.M.S.S. and Silva, J.C. (1987) Water Hyacinth (<u>Eichhornia crassipes</u>) in Ruminant Nutrition. <u>Wat. Sci. Tech.</u>, Vol.19 (10), 109-112

Abernathy, A.R., Zirschky, J. and Brett, B.M. (1985) Overland Flow Wastewater Treatment at Easley. S.C. J.W.P.C.F., Vol.57 (4), 291-299

Adams, F.S., Cole, H. JR and Massier, L.B. (1973) Element Constitution of Selected Aquatic Vascular Plants from Pennsylvania: Submersed and Floating Leaved Species and Rooted Emergent Species. <u>Environ. Pollut.</u>, Vol.5 (2), 117-147

Adriano, D.C., Paulsen, G.M. and Murphy, L.S. (1971) Phosphorus-iron and Phosphorus-zinc Relationshipsin Corn (<u>Zea mays L.</u>) Seedlings as Affected by Mineral Nutrition. <u>Agron. J.</u>, Vol.63, 36-39

Alexander, W.V. (1985) Phosphate Removal from Effluents. Presented Symposium on the Impact of Phosphate on South African Waters. Pretoria.

Alexander, W.V. and Wood, A. (1987) Experimental Investigations into the Use of Emergent Plants to Treat Sewage in South Africa. <u>Wat. Sci.</u>, Vol.19 (10), 51-59

Allus, M.A., Brereton, R.G. and Nickless, G (1988) Chemometric Studies of the Effect of Toxic Metals on Plants: The Use of Response Surface Methodology to

- 121 -

Investigate the Influence of Tl, Cd and Ag on the Growth of Cabbage Seedlings. <u>Environ. Pollut.</u>, Vol.52 (3), 169-181

Antonovics, J. (1975) Metal Tolerance in Plants: Perfecting an Evolutionary Paradigm. In: Hutchinson, T.C., Epstein, S. Page, A.L., Van Loon, J and Davey, T. (Edits) International Conference on <u>Heavy Metals in</u> <u>Environment</u>, Vol.II (1), 169-186

Arthur, R.A.J. (1986) At the Root of the Matter. <u>Water &</u> <u>Wastewater Treatment</u>, Vol.3, 43-48

Van Assche, F., Clijsters, H. and Marcelle, R. (1979) Photosynthesis in <u>Phaseolus vulgaris</u> as Influenced by Supra-optimal Zinc Nutrition. In: Marcelle, R., Clijsters, H. and Van Poucke, M., (Edits) <u>Photosythesis</u> and Plant Development, The Hague, Junk, 175-84

Van Assche, F., Cardinaels, C. and Clijsters, H. (1988) Induction of Enzyme Capacity in Plants as a Result of Heavy Metal Toxicity: Dose - Response Relations in <u>Phaseolus vulgaris L.</u> Treated with Zinc and Cadmium. <u>Environ. pollut.</u>, Vol.52, 103-115

Aulio, K. (1980) Accumulation of Copper in Fluvial Sediments and Yellow Water Lilies (<u>Nuphar lutea</u>) at Varying Distances From a Metal Processing Plant. <u>Bull.</u> <u>Environ. Contam. Toxicol.</u>, Vol.25, 713-717

Baker, A.J.M. (1987) Metal Tolerance. <u>New Phytologist</u>, Vol.106, 93-111 Baker, A.J.M. and Walker, P.L. (1989) Physiological Responses of Plants to Heavy Metals and the Quantification of Tolerance and Toxicity. <u>Chemical</u> <u>Speciation and Bioavailability</u>, Vol.1, 7-17

Baker, A.J.M. and Brooks, R.R. (1989) Terrestrial Higher plants which Hyper-accumulate Metallic Elements - A Review of their Distribution, Ecology and Phytochemistry. <u>Biorecovery.</u> Vol.1, 81-126

Baker, A.J.M. and Walker, P.L. (1990) Ecophysiology of Metal Uptake by Tolerant Plants. In: Shaw, A.J.M. (Edit) <u>Heavy Metal Tolerance in Plants, Evolutionary Aspects</u>, CRC Press, Florida, 155-177

Baker, J.M. (1983) Impact of Oil Pollution on Living Resources. Commission on Ecology Papers Number 4 IUCN, 13-30

Bates, R.P. and Hentges, J.F. (1976) Aquatic Weeds-Eradicate or Cultivate. Econ. Botany, Vol.3, 39-50

Bayes, C.D., Bache, D.H. and Dickson R.A. (1989) Land -Treatment Systems: Design and Performance with Special Reference to Reed Beds. <u>J. Inst. Wat. Eng. Man.</u>, Vol.3, 588-598

Beeftink, W.G., Nieuwenhuize, J., Stoeppler, M. and Mohl, M. (1982) Heavy Metal Accumulation in Salt Marshes from the Western and Easten Scheldt. <u>Sci. Total Environ.</u> Vol.25, 199-223 Black, G. (1979) Decomposition of Macrophytes with Uniformly C 14 Labelled Plant Material. <u>Arch. Hydrobiol.</u>, Vol.85 (3), 326-335

Black, G. and DuBois, J.P. (1982) L'epuration des eaux par les plantes aquatiques. <u>AFEE.</u>, 103 pp

Black, G., Gagnaire-Michard, J. and Kirassian, B. (1984) Accumulation du Zinc (Zinc 65) par une Macrophyte: <u>Typha</u> <u>latifolia</u>. <u>Sc. de l'eau</u>, Vol.3, 241-248

Black, R.A.P. and Mitchell, R.L. (1952) Trace Elements in the Common Brown Algae and in Seawater. <u>J. Mar. Biol.</u> <u>Ass. U.K.</u>, Vol.30, 575-584

Blake G., Kaigate B., Fourcy, A. and Boutin, C. (1987) Incorporation of Cadmium by Water Hyacinth. <u>Wat. Sci.</u> <u>Technol.</u>, Vol.19 (10), 123-128

Blake, G., Gagnaire-Michard, J., Kirassian, B. and Morand, P. (1987) Distribution and Accumulation of Zinc in <u>Typha latifolia</u>. In: <u>Proc. Seminar on Aquatic Plants</u> for Wat. Treat. and Res. Recov. Orlando, Florida, 487-495

Boardman, N.K. (1975) Trace Elements in Photosynthesis. In: Egan, D.J.D. and Egan, A.R. (Edits) <u>Trace Elements in</u> <u>Soil-Plant-Animal Systems</u>, Academic Press, 199-212

Boutin, C. (1987) Domestic Wastewater Treatment in Tanks Planted with Rooted Macrophytes: Case Study; Description of the System; Design Criteria; and Efficiency. <u>Wat. Sci.</u> Technol., Vol.19 (10), 29-40 Boyd, C.E. (1970a) Production, Mineral Accumulation and Pigment Concentrations in <u>Typha latifolia</u> and <u>Scirpus</u> <u>americanus</u>. <u>Ecology</u>, Vol.51, 285-290

Bryan, G.W. and Hummerstone, L.G. (1973) Brown Seaweeds as Indicators of Heavy Metals in Estuaries in South-West England. J. Mar. Biol Ass. U.K., Vol.53, 705-720

Bryce-Smith, D. (1975) Heavy Metals as Contaminants of Human Environment. <u>The Educational Techniques Subject</u> Group, Chemistry Cassette, The Chem. Soc., London

Bucksteeg, K. (1987) Sewage Treatment in Helophyte Beds-First Experiences with a New Treatment Procedure. <u>Wat.</u> Sci. Technol., Vol.19 (10), 1-10

Carlson, R.W., Bazzaz, F.A. and Rolfe, G.L. (1975) The Effect of Heavy Metals on Plants. II. Net Photosynthesis and Transpiration of Whole Corn and Sunflower Plants Treated with Pb, Cd, Ni and Ti. <u>Environ. Res.</u>, Vol.10, 113-20

Carpenter, K. (1924) A Study of the Fauna of Rivers Polluted by Lead Mining in the Aberystwyth District of Cardiganshire. <u>Ann. App. Biol.</u>, Vol.11, 1-23

Carpenter, S.R. (1980) Enrichment of Lake Wingra, Wisconsin, by Submerged Macrophyte Decay. <u>Ecology</u>, Vol.61, 1145-55

Carpenter, S.R. (1981) Submerged Vegetation: An Internal Factor in Lake Ecosystem Succession. <u>Am. Nat.</u>, Vol.118, 372-83 Chigbo, F.E., Smith, R.W. and Shore, F.L. (1982) Uptake of Arsenic, Cadmium, Lead and Mercury from Polluted Waters by the Water Hyacinth <u>Eichhornia crassipes</u>. <u>Environ. Pollut.</u> (Series A), Vol.27, 31-36

Clijsters, H. and Van Assche, F. (1985) Inhibition of Photosynthesis by Heavy Metals. <u>Photosynth. Res.</u>, Vol.7, 31-40

Cloris, M. and Araujo, H. (1987) Use of Water Hyacinth in Tertiary Treatment of Domestic Sewage. <u>Wat. Sci.</u> <u>Technol.</u>, Vol.19 (10), 11-17

Cooper, P.F., Hobson, J.A. and Jones, S. (1989) Sewage Treatment by Reed Bed Systems. <u>J. Inst. Wat. Eng. Man.</u>, Vol.3, 60-74

Cottenie, A., Dhaese, A. and Cametrlynck, R. (1976) Plant Quality Response to Uptake of Polluting Elements. <u>Fds.</u> <u>Hum. Nutr.</u>, XXVI 1/3, 293-319

Daniels, R.R., Stuckmeyer, B.E. and Peterson, L.A. (1972) Copper Toxicity in <u>Phaseolus vulgaris L.</u> as Influenced by Iron Nutrition. I. An Anatomical Study. <u>J. Amer. Soc.</u> Hort. Sci., Vol.9, 249-254

Davies, B.E. and Ginnever, R.C. (1979) Trace Metal Contamination of Soil and Vegetables in Shipham Somerset. J. Agric. Sci. Camb., Vol.93, 753-756

DeBusk, T.A. and Reddy, K.R. (1990) Use of Floating Macrophytes for Wastewater Treatment. IAWPRC Specialist Group: The Use of Macrophytes in Water Pollution Control, <u>Newsletter</u>, Vol.3, 66-70 DeLaune, R.D., Smith, C.J., Patrick, W.H., Fleeger, J.W. and Tolley, M.D. (1984) Effect of Oil on Salt Marsh Biota: Methods for Restoration. <u>Environ. Pollut.</u> (Ser. A), Vol.36, 207-228

Dicks, B. and Hartley, J.P. (1982) The Effects of Repeated Small Oil Spillages and Chronic Discharges. Phil. Trans. R. Soc. Lond, Vol.B

Dryssen, D.C., Patterson, J.V. and Weichart, G.F. (1971) Inorganic Chemicals. In: Proceedings of FAO Conference on Marine Pollution and its Effects on Living Resources and Fishing, Rome, 1970

Dykyjova, D. and Kvet, J. (1978) Pond Littoral Ecosystems. Springer-Verlag, New York, USA.

Eaton, A. (1979) Leachable Trace Elements in San Francisco Bay Sediments: Indicators of Sources and Estuarine Processes. <u>Environ. Geol.</u>, Vol.2 (6), 333-339

Ellis, J.B. (1976) Sediments and Water quality of urban storm water. <u>Water Services</u>, Vol.80 (970), 730-734

Ellis, J.B. and Harrop, D.O. (1984) Variations in Solids Loadings to Roadside Gully Pots. <u>Sci. Total Environ.</u>, Vol.33, 203-211

Ellis, J.B., Revitt, D.M., Shutes, R.B.E. and Hamilton, R.S. (1986) The Effects of Urbanisation on Receiving Water Quality: Heavy Metal Toxicity. In: Solbe, J. (Edit) <u>The Effects of Land Use on Fresh Waters</u>, Ellis Horwood Ltd, Chichester, 473-477 Ellis, J.B. (1986) Pollution Aspects of Urban Runoff. In: Torno, H.C., Marsalek, J. and Desbordes, M. (Edits) NATO ASI Series, <u>Urban Runoff Pollution</u>, Vol.G10, Springer-Verlag, Berlin, Heidelberg.

Farago, M.E., Clark, A.J. and Pitt, M.J. (1975) The Chemistry of Plants which Accumulate Metals. <u>Coordination</u> <u>Chemistry Reviews</u>, Vol.16, 1-8

Foght, J.M. and Westlake, D.W.S. (1987) Biodegradation of Hydrocarbons in Freshwater Oil in Freshwater. <u>Chemistry</u>, <u>Biology</u>, <u>Countermeasure Technology</u>, 217-230

Forstner, W. and Wittman, G.T.W. (1979) Metal Pollution in the Aquatic Environment. Springer-Verlag New York.

Forstner, W. and Wittmann, G.T.W. (1981) Metal Pollution in the Aquatic Environment. 2nd Edition, Springer-Verlag, Berlin.

Frederiksen, O.T. (1987) The Fight Against Eutrophication in the Inlet of 'Odense Fjord' by Reaping of Sea Lettuce (<u>Ulva lactuca</u>). <u>Wat. Sci Techno</u>l., Vol.19 (10), 81-87

Friberg, L., Piscator, M., Nordberg, G.F. and Kiellstrom, T. (1972) Cadmium in the Environment. <u>Critical Reviews of</u> Chemistry.

Gavens, A., Revitt, D.M. and Ellis, J.B. (1982) Hydrocarbon Accumulation in Freshwater Sediments of an Urban Catchment. <u>Hydrobiologia</u>, Vol.91, 285-292 GLC, (1986) A Nature Conservation Strategy for London: Woodland, Wasteland, the Tidal Thames and Two London Boroughs. <u>Ecology Handbook</u>, No.4

Grace, J.B. (1984) Effects of Tubificid Worms on the Germination and Establishment of <u>Typha</u>. <u>Ecology</u>, Vol.65, 1689-1693

Grace, J.B. and Harrison, J.S. (1986) The Biology of Canadian Weeds, <u>Typha angustifolia</u>, <u>Typha latifolia</u>, and <u>Typha glauca</u>. <u>Can. J. Plant Sci</u>., Vol.66, 361-379

Grace, J.B. (1987) The Impact of Preemption on the Zonation of Two <u>Typha</u> Species along Lakeshores. <u>Ecology</u> <u>Monographs</u>, Vol.57 (4), 283-303

Grahn, O. and Hakanson, L. (1987) Case Study on Metal Distribution and Uptake in Biota. In: Landner, L. (Edit) <u>Speciation of Metals in Water, Sediment and Soil Systems</u>, Springer-Verlag, Berlin, 127-144

Guilizzoni, P. (1975) Manganese, Copper and Chromium Content in Macrophytes of Lake Endine (Northern Italy). <u>Mem. Ist. Ital. Idrobiol. Dot, Marco de Marchi</u>, Vol.32, 313-332

Gupta, G. (1980) Use of Water Hyacinths in Wastewater Treatment. J. Environ. Health, Vol.43 (2), 80-82.

Hakanson, L. (1980) An Ecological Risk Index for Aquatic Pollution Control - A Sedimentological Approach. <u>Wat.</u> <u>Res.</u>, Vol.14, 975-1001 Hall, M.J. (1977) The Effect of Urbanisation on Storm Runoff from Two Cachment Areas in NW London. Symposium Publication 123, IAHS-UNESCO, Amsterdam, 144-152

Hall, M.J., Hockin, D.L. and Ellis, J.B. (1990) The Design of Flood Storage Reservoirs. Butterworths, London in press.

Hardy, J.K. and O'Keeffe, D.H. (1985) Cadmium Uptake by the Water Hyacinth: Effects of Root Mass, Solution Volume, Complexes and other Metal Ions. <u>Chemosphere</u>, Vol.14 (5), 417-426.

Harrison, R.M. (1979) Toxic Metals in Street and Household dusts. <u>Sci. Total Environ.</u>, Vol.11, 89-97

Hardy, J. K. and Raber, N.B. (1985) Zinc Uptake by the Water Hyacinth, Effects of Solution Factors. <u>Chemosphere</u>, Vol.14 (9), 1155-1166

Harrison, R.M., Laxen, P.H. and Wilson, S.J. (1981) Chemical Associations of Lead, Cadmium, Copper, and Zinc in Street Dusts and Roadside Soils. <u>Environ. Sci.</u> <u>Technol.</u>, Vol.15 (11), 1378-1383

Harrison, R.M. and Laxen, D.P.H. (1981) Lead Pollution : Causes and Control. Chapman and Hall, London

Harrison, R. M. and Chirgawi, M. B. (1989) The Assessment of Air and Soil as Contributors of Some Trace Metals to Vegetable Plants I. Use of a Filtered Air Growth Cabinet. <u>Sci. Total Environ.</u>, Vol.83, 13-34
Harrison, R. M and Chirgawi, M.B. (1989) The Assessment of Air and Soil as Contributors of Some Trace Metals to Vegetable Plants II. Translocation of Atmospheric and Laboratory-Generated Cadmium Aerosols to and within Vegetable Plants. <u>Sci. Total Environ.</u>, Vol.83, 35-45

Harrison, R.M. and Chirgawi, M.B. (1989) The Assessment of Air and Soil as Contributors of Some Trace Metals to Vegetable Plants III. Experiments with Field-Grown Plants. <u>Sci. Total Environ.</u>, Vol.83, 47-62

Haslam, S.M. and Wolseley, P.A. (1981) River Vegetation: Its Identification Assessment and Management. A Field Guide to the Macrophytic Vegetation of British Water Courses. Cambridge University Press, 160-161

Haslam, S.M. (1982) A Proposed Method For Monitoring River Pollution Using Macrophytes. <u>Environ. Technol.</u> <u>Lett.</u>, Vol.3, 19-34

Haslam, S.M. (1987) River Plants. Cambridge University Press.

Heisey, R.M. and Damman, A.W.H. (1982) Copper and Lead Uptake by Aquatic Macrophytes in Eastern Conneticut, USA. Aquat. Bot., Vol.14, 213-229

Howard-Williams, C. and Davies, B.R. (1979) The Ratio of Dry Matter and Nutrient Loss from Decomposing <u>Potamogeton</u> <u>pectinatus</u> in a Brackish South Temperate Lake. <u>Freshwat</u>. <u>Biol.</u>, Vol.9, 13-21 Jaworski, J.F. (1979) Effects of Lead in the Environment - 1978, Quantitative Aspects. National Research Council of Canada, Ottawa, NRCC 16736

Jewell, W.J. (1971) Aquatic Weed Decay: Dissolved Oxygen Utilization and Nitrogen and Phosphorus Regeneration. J. <u>Wat. Pollut. Control Fed.</u>, Vol.43, 1457-1467

John, C.K. (1984) Treatment of Agro-industrial Wastes Using Water Hyacinth. <u>Wat. Sci. Technol.</u>, Vol.17, 781-790

Jones, J.R.E. (1935) The Toxic Action of Heavy Metal Salts on the Three Spined Stickleback (<u>Gasterosteus</u> <u>aculeatus</u>). <u>J. Expt. Biol</u>., Vol.12, 165-73

Jyung, W.H., Ehmann, A., Schlender, K.K. and Scala, J. (1975) Zinc Nutrition and Starch Metabolism in <u>Phaseolus</u> <u>vulgaris L.</u>. <u>Plant Physiol</u>., Vol.55, 414-420

Kelly, P.C., Brooks, R.R., Dilli, S. and Jaffré, T. (1975) Preliminary Observations on the Ecology and Plant Chemistry of some Nickel-accumulating Plants from New Caledonia. <u>Proc. Royal Soc. Lond.</u>, Section B, Vol.189, 69-80

Kersten, W.J., Brooks, R.R., Reeves, R.D. and Jaffre, T. (1980) Nature of Nickel Complexes in <u>Psychotria douarrei</u> and other Nickel-accumulating Plants. <u>Phytochemistry</u>, Vol.19, 1963-1965

Kiesling, R.W., Alexander, S.K. and Webb, J.W. (1988) Evaluation of Alternative Oil Spill Cleanup Techniques in a <u>Spartina alternifolora</u> Salt Marsh. <u>Environ. Pollut.</u>, Vol.55, 221-238 Kickuth, R. and Kaitzis, G. (1975) Mikrobizid wirksame aromaten aus <u>Scirpus lacustris</u>. Forum Umwelltyg, 4/5, 134 PP

Kickuth, R. (1980) Abwasserreinigung in Mosaikmatrizen aus anaeroben und aeroben Teilbezirken. <u>Schriftenreihe</u> <u>Wasser Abwasser</u>, 19, Bd. 2, 639-665

Kickuth, R. (1984) Das Wurzelraumverfahren in der Praxis. Landschaft und Stadt, Vol.16 (3), 145-153

Kobayshi, J. (1971) Relation between the 'Itai-Itai' Disease and the Pollution of River Water by Cadmium from a Mine. In: Proceedings of the 5th International Conference, <u>Advances in Water Pollution Research</u>, San Francisco, 1971, Hawaii I-25, 1-7

Kovalskii, V.V., Grivovskaya. I.F. and Samarina, B.F. (1976) Concentration of Trace Elements by Aquatic Plants. In: Proceedings of Lake Biology All-Union Symposium on <u>Basic Problems of Fresh-Water Lakes</u> [in Russian], Vol.3, 79-86

Krusi, B.O. and Wein, R.W. (1988) Experimental Studies on the Resiliency of Floating <u>Typha</u> mats in a Freshwater Marsh. J. Ecol., Vol.76, 60-72

Kyte, D. (1984) In: Rivers and Wildlife Handbook. Royal Society for the Protection of Birds, Royal Society for Nature Conservation.

Lagerwerff, J.V. and Brower, D.L. (1975) Source Determination of Heavy Metal Contaminants in the Soil at a Mine and Smelter Area. In: Hemphill, D.D. (Edit) <u>Trace</u> Substances in Environmental Health, Vol.9, Univ. Missouri Press, 207-216

Lan, C., Chen, G., Li, L. and Wong, M.H. (1990) Purification of Wastewater from a Pb/Zn Mine Using Hydrophytes. In: Cooper, P.F.and Pindlater, B.C. (Edits) Proceedings of International Conference on <u>Use of</u> Constructed Wetlands in Water Pollution Control, 451-459

Larsen, V. and Schierup, H. (1981) Macrophytes Cycling of Zinc, Copper, Lead and Cadmium in the Littoral Zone of a Polluted and a Non-polluted Lake II: Seasonal Changes in Heavy Metal Content Above-ground Biomass and Decomposing Leaves of <u>Phragmites australis Trin.</u> <u>Aquatic Botany</u>, Vol.11, 211-230

Lawson, G. T. (1985) Cultivating Reeds (<u>Phragmites</u> <u>australis</u>) for Root-Zone Treatment of Sewage. Conf. Rep., W.R.C. ITE Project 965

Lee, J.R., Brooks, R.R. and Jaffré, T. (1977b) Isolation and Identification of Citric Acid in some Nickelaccumulating Plants. <u>Phytochemistry</u>, Vol.16, 1503-1505

Leland, H.V. and McNurney. J.M. (1984) Lead Transport in a River Ecosystem. In: Proc. International Conf. <u>Transport of Persistent Chemicals in Aquatic Ecosystems</u>, Ottawa, 3-17-23

Lepp, N.W. (Edit) (1981) <u>The Effect of Heavy Metals on</u> <u>Plants</u>. London, Applied Science Publishers. Loehr, R.C., Jewell, W.J., Novak, J.D., Clarkson, W.W. and Friedman, G.S. (1979) Land Application of Wastes. Vols.1 and 2, Van Nostrand Reinhold, New York.

Luoma, S.N. and Davis, J.A. (1983) Requirements for Modelling Trace Metal Partitioning in Oxidised Estuarine Sediments. <u>Mar. Chem.</u>, Vol.12, 159-181

Mance, G. (1987) Pollution Threat of Heavy Metals in the Aquatic Environment. Elsevier Applied Science, London.

Marschner, H. (1986) Mineral Nutrition of Higher Plants. Academic Press, London

May, T.W. and Mckinney, G.L. (1981) Cadmium, Lead, Mercury, Arsenic and Selenium Concentration in Freshwater Fish, 1976-1977. National Pesticide Monitoring Program. J. Pestic. Monit., Vol. 15, 14-38

McNaughton, S.J., Folsom, T.C., Lee, T., Park, F., Price, C., Roeder, D., Schmitz, J. and Stockwell, C. (1974) Heavy Metal Tolerance in <u>Typha latifolia</u> without the Evolution of Tolerant Races. <u>Ecology</u>, Vol.55, 1163-1165

McNaughton, S.J. (1975) r- and k- selection in <u>Typha</u>. <u>American Naturalist</u>, Vol.109, 251-261

Mengel, K. and Kirkby, E.A. (Edits) (1978) Principles of Plant Nutrition. International Potash Institute.

Merchyulenene, D.P. and Nyanishkene V.B. (1976) Accumulation of Lead-210 by Fresh Water Plants. <u>Radiobiol</u>, Vol.16, 200-204 Middlebrooks, E.J. and Reed, S.C. (1981) The Flowering of Wastewater Treatment. <u>J. Wat. Eng. Man.</u>, Vol.6, 50-54

Miller, J.E., Hassett, J.J. and Koeppe, D.E. (1979) Interaction of Pb and Cd on Metal Uptake and Growth of Corn Plants. <u>J. Environ. Qual.</u>, Vol.6, 18-20

Moore, J.W. and Ramamoorthy, S. (1984) Heavy Metals in Natural Waters; Applied Monitoring and Impact Assessment. New York, Springer-Verlag.

Moorhead, K.K. and Reddy, K.R. (1988) Oxygen Transport through Selected Aquatic Macrophytes. <u>J. Environ. Qual.</u>, Vol.17 (1), 138-142

Morozov, N.V. (1984) Use of Macrophytes for Treating Agricultural Runoff. <u>Water Resources</u>, Vol.11, 259-268

Morrison, G.M.P., Revitt, D.M., Ellis, J.B., Balmer, P., and Svensson, P. (1984) The Physico-Chemical Speciation of Zn, Cd, Pb and Cu in Urban Stormwater. 989-1000 In: Balmer P, Malmquist P-A and Sjoberg A. (Edits) <u>Planning</u> <u>and Control of Urban Storm Drainage</u>, Chalmers, University, Gothenburg, Sweden.

Morrison, R.S., Brooks, R.R. Reeves, R.D. and Malaisse, F. (1979) Copper and Cobalt Uptake by Metallophytes from Zaire. <u>Plant and Soil</u>, Vol.53, 535-539

Mortimer, D.C. (1985) Freshwater Aquatic Macrophytes as Heavy Metal Monitors - The Ottawa River Experience. Environ. Mon. Assess., Vol.5, 311-323 Mudroch, A. (1980) Biogeochemical Investigation of Big Creek Marsh, Lake Erie, Ontario. <u>J. Great Lakes Res.</u>, Vol. 6, 338-347

Mudroch, A. and Capobianco, J. (1978) Study of Selected Metals in Marshes on Lake St. Clair, Ontario. <u>Arch.</u> <u>Hydrobiol.</u>, Vol.84, 87-108

Muramoto, S. and Oki, Y. (1983) Removal of Some Heavy Metals From Polluted Water by Water Hyacinth (<u>Eichhornia</u> <u>crassipes</u>). <u>Bull. Environ. Contam. Toxicol.</u>, Vol.30, 170-177

Nasu, Y. and Kugimoto, M. (1983) Comparative Studies on the Absorption of Cadmium and Copper in <u>Lemna</u> <u>paucicostata</u>. <u>Environ. Pollut</u>. (Series A), Vol.32, 201-209

Oliver, B.G. (1973) Heavy Metal Levels of Ottawa and Rideau River Sediment. <u>Environ. Sci. Technol.</u>, Vol.7 (2), 135-137

Orekhovskii, A.R. (1986) Use of Reed and Bulrush for Biological Stabilization of the Underwater Slopes of Sand Embankments. <u>Lesovod. Agromelior</u>, Vol.1, 114-125

Oron, G.L.R. and Wildschut, D. (1984) Wastewater Recycling by Duckweed for Protein Production and Effluent Renovation. <u>Wat. Sci. Technol.</u>, Vol.17 (4/5), 803-817

Overcash, M.R. and Pal, D. (1976) Design and Land Treatment Systems for Industrial Waters: Theory and Practice. Ann Arbor Science, Michigan. Pinto, C.L.R., Caconia, A. and Souza, M.M. (1987) Utilization of Water Hyacinth for Removal and Recovery of Silver from Industrial Wastewater. <u>Wat. Sci. Technol.</u>, Vol.19 (10), 89-101

Prahalad, A.K. and Seenayya, G. (1989) Physico-chemical Interactions and Bioconcentration of Zinc and Lead in the Industrially Polluted Husainsagar Lake, Hyderabad, India. Environ. Pollut., Vol.58, 139-154

Preston, A., Jeffries, D.F., Dutton, J.W.R., Harvey, B.R. and Steele, A.K. (1972) British Isles Costal Water: The Concentration of Selected Heavy Metals in Seawater, Suspended matter and Biological Indicators - a Pilot Survey. <u>Eviron. Pollut.</u>, Vol.3, 69-82

Price, C.A., Clark, H.E. and Funkhouser, H.E. (1972) Functions of Micronutrients in Plants. <u>Soil Sci. Soc. of</u> <u>America</u>, Madison/Wisconsin, 731-742

Ray, S. and White, W. (1976), <u>J. Environ. Sci. Health</u>, A-11, 717 pp

Reed, R.H. and Darrig, S. (1983) Physiological Responses of Ship-fouling and Non-fouling Isolates of <u>Enteromorpha</u> <u>compressa</u> to Copper. In: <u>Proc. Inter. Conf. Heavy Metals</u> <u>Environ.</u> CEP Consultants, Heidelberg, 322-325

Reimer, P. (1989) Concentrations of Lead in Aquatic Macrophytes from Shoal Lake, Manitoba Canada. <u>Eviron.</u> <u>Pollut.</u>, Vol.56, 77-84

Reimer, D.N. and Toth, S.J. (1968) A Survey of the Chemical Composition of Aquatic Plants in New Jersey. <u>New</u> Jersey Agric. Exp. Stn. Coll. Agric. Environ. Sci., Rutgers University Bull. 820. (As Cited in Hutchinson (1975).)

Revitt, D.M. and Ellis, J.B. (1980) Rain Water Leachates of Heavy Metals in Road Surface Sediments. <u>Wat. Res.</u>, Vol.14, 1403-1407

Revitt, D.M. and Morrison, G.M. (1987) Metal Speciation Variations with Separate Stormwater Systems. <u>Eviron.</u> <u>Technol. Lett.</u>, Vol.8, 373-380

Riley, G.A. (1970) Correlation of Organic Matter in Sea Water. Advances in Marine Biology, Vol.8, 1-118

Romero, F., Elejalde, C. and Gomez, G. (1989) Monitoring of Plant and Soil Pollution Based on Ionic Impulsion. Wat. Air Soil Pollut., Vol.44, 295-306

Salomons, W. (1985) Sediment and Water Quality. <u>Environ</u>. Technol. Lett., Vol.6, 315-326

Santos, E.J., Silva, E.H.B.C., Fiuza, J.M. and Batista, T.R.O. (1987) A High Organic Load Stabilization Pond Using Water Hyacinth - A 'Bahia' Experience. <u>Wat. Sci.</u> Technol., Vol.19 (10), 25-28

Sartor, J.D. and Boyd, G.B. (1972) USEPA, Report, EPA-R2-72-081

Seeliger, U. and Edwards, P. (1977) Correlation Coefficients and Concentration Factors of Copper and Lead in Seawater and Benthic Algae. <u>Mar. Pollut. Bull.</u> Vol.8 (1), 16-19 Seeliger, U. (1979) Fate of Biologically Accumulated Copper in Growing and Decomposing Thalli of Two Benthic Red Marine Algae. <u>J. Mar. Biol. Ass. U.K.</u>, Vol.59, 227-238

Seeliger, U. and Cordazzo, С. (1982)Field and Experimental Evaluation of Enteromorpha sp. as а Quantitative Monitoring Organism for Copper and Mercury in Estuaries. Environ. Pollut. (Series A), Vol.20, 197-206

Seeliger, U. and Lacerda, L.D. (1986) Uptake and Release of Copper and Mercury by the Alga <u>Ceramium brasiliense</u>. Revta Brasil. Bot., Vol.9, 63-68

Seidel, K. (1966) Reinigung von Gewassern durch hohere Pflanzen. <u>Deutsche Naturwissenschaft</u>, Vol.12, 289-297

Seidel, K. (1976) Macrophytes and Water Purification. In: Tourbier, J. and Pierson, R.W. (Edits) <u>Biological Control</u> <u>of Water Pollution</u>, University of Pennsylvania, Pennsylvania.

Seidel, K. and Happel, H. (1978) Contributions to Revitalisation of Waters: Stiftung Limnologische Arbeitsgrupe. Meersburg, West Germany.

Seidel, K. and Happel, H. (1983) Limnologie in Stichworten (IV). Kleine Pflanzen-Klaranlagen. Wasserkalender, 119-141

Shelef, G. and Soeder, C.J. (1980) (Edits). Algae Biomass Production and Use. Elsevier-North Holland Press. Shutes, R.B.E. (1985) The Invertebrates of the Dew Pond and Icehouse Wood, Trent Park. Middlesex Polytechnic Press.

Shutes, R.B.E. (1985) Trent Park Natural History Guide. Middlesex Polytechnic Press.

Simeon, C. and Silhol, M. (1987) Purification of Pisciculture Waters Through Cultivation and Harvesting of Aquatic Biomass. <u>Wat. Sci. Technol.</u>, Vol.19.(10), 113-121

Sridhar, M.K.C. (1986) Trace Element Composition of <u>Pistia stratiotes L.</u> in a Polluted Lake in Nigeria. <u>Hydrobiologia</u>, Vol.131, 273-2756

Sturges, W.T. and Harrison, R.M. (1985) An assessment of the Contribution from Paint Flakes to the Lead Content of some Street and House dust. <u>Sci. Tot. Environ.</u>, Vol.44, 225-234

Sundblag, K and Wittgren, H.B. (1989) <u>Glyceria maxima</u> for Wastewater Nutrient Removal and Forage Production. <u>Biol.</u> <u>Waste</u>, Vol.27, 29-42

Taylor, G.J. and Crowder, A.A. (1983) Accumulation of Atmospherically deposited Metals in Wetland Soils of Sudbury, Ontario. <u>Wat. Air. Soil Pollut.</u>, Vol.19 (1), 29-42

Taylor, G.J. and Crowder, A.A. (1983) Uptake and Accumulation of Heavy Metals by <u>Typha latifolia</u> in Wetlands of the Sudbury, Ontario Region. <u>Can. J. Bot.</u>, Vol.61, 63-73 Tessier, A. and Campbell, P.G.C. (1987) Partitioning of Trace Metals in Sediments: Relationships with Bioavailability. <u>Hydrobiologia</u>, Vol.149, 43-52

Thurman, D. and Hardwick, K. (1988) How Plants Survive an Overdose of Metal. <u>New Scientist</u>, Vol.10, 44-45

Tokunaga, T. Furuta, N and Morimoto, M. (1976) Accumulation of Cadmium in <u>Eichhornia crassipes</u> Solms. <u>J.</u> <u>Hyg. Chem.</u>, Vol.22, 234-9

Trivedy, P.K. and Gudekar, V.R. (1987) Treatment of Textile Industry Waste Using Water Hyacinth. <u>Wat. Sci.</u> Technol., Vol.19 (10), 103-107

Tsutomu, I. and Teranish, S. (1988) Rates of Nutrient Uptake and Growth of the Water Hyacinth (<u>Eichhornia</u> <u>crassipes</u> (MART.) Solms). <u>Wat. Res.</u>, Vol.22 (8), 943-951.

USEPA (1980) Ambient Water Quality Criteria for Lead. EPA-440/5-80-057, Washington, DC.

USEPA (1977) Process Design Manual for Land Treatment of Municipal Wastewater. EPA-625/1-77-008, Washington, DC.

Valk, V.D. and Davis, C.B. (1976) The Seed Banks of Prairie Glacial Marshes. <u>Can. J. Bot.</u>, Vol.54, 1832-1838

Van Hassel, J.H., Ney, J.J. and Garling, D.L. (1979) Seasonal Variations in the Heavy Metal Concentrations of Sediments Influenced by Highway of Different Traffic Volumes. <u>Bull. Environ. Contam. Toxicol.</u>, Vol.23, 592-596 Varenko, N.I. and Chuiko, V.T. (1971) Role of Higher Aquatic Plants in the Dneprodzerzhinsk Reservoir. <u>J.</u> <u>Hydrobiol.</u>, Vol.7, 45-48 (As Cited in Hutchinson (1975).)

Wainwright, S.J. and Woolhouse, H.W. (1975) Physiological Mechanisms of Heavy Metal Tolerance Plants. In: Chadwick, M.J. and Goodman, G.T. (Edits) The Ecology of Resource Degradation and Renewal, Blackwell, Oxford

Wang, W. (1986) Toxicity Tests of Aquatic Pollutants by Using Common Duckweed. <u>Environ. Pollut.</u> (Series B), Vol.11, 1-14

Weber, S.A. and Tchobarioglous, G. (1985) Rational Design Parameters for Ammonia Conversion in Water Hyacinth Treatment Systems. <u>J.S.P.D.F.</u>, Vol.57 (4), 316-323

Welsh, R.P.H. and Denny P. (1980) The Uptake of Lead and Copper by Submerged Aquatic Macrophytes in Two English Lakes. J. Ecol., Vol.68, 443-55

Van der Werff, M. and Pruyt, M.J. (1982) Long-term Effects of Heavy Metals on Aquatic Plants. <u>Chemosphere</u>, Vol.11, 727-739

WHO (1989) Lead - Environmental Aspects. Environmental Health Criteria 85, WHO, Geneva.

Werner, M.D. and Adams, V.D. (1984) Consequences of Oil Pollution on the Decomposition of Vascular Plant in Litter in Freshwater Lakes: Part 2 - Nutrient Exchange Between Litter and the Environment. <u>Environ. Pollut.</u> (Series A), Vol.34, 101-117 Widyanto, L.S., Sopannata, A. and Fried, U.S. (1983) Water Hyacinth as A Potential Plant in A paper Factory. J. Aquat. Plant Man.., Vol.21, 32-35.

Winter, M. and Kickuth R. (1989) Elimination of Sulphur Compounds from Wastewater by The Root Zone Process-I. Performance of a Large-Scale Purification Plant at a Textile Finishing Industry. <u>Wat. Res.</u>, Vol.23 (5), 535-546

Wolverton, B.C. and McDonald, R.C. (1975) Compiled Data on the Vascular Aquatic Plant Program. Bay St. Louis, Miss, NASA, National Technology Laboratories, 115 pp

Wolverton, B.C. and McDonald, R.C. (1975) Application of Vascular Aquatic Plants for Pollution Removal, Energy and Food Production in a Biological System. Bay St Louis, Miss, NASA, National Space Technology Laboratories, 150 PP

Wolverton, B.C. and McDonald, R.C. (1979) The Water Hyacinths from Prolific Peat to Potential Provider. <u>Ambio</u>, Vol.8 (1), 2-9

Xian, X. (1989) Response of Kidney Bean to Concentration and Chemical Form of Cadmium, Zinc, and Lead in Polluted Soils. <u>Environ. Pollut.</u>, Vol.57, 127-137

Xie, X., Zhang, Y.R. and Zhang, T.T. (1984) Purifying Sewage Water with Water Hyacinth. <u>Environ. Sci.</u>, Vol.5 (3), 18-25 Yamamoto, T. (1972) The Relation between Concentration Factors in Seaweeds and Residence Time of Some Elements in Seawater. Rec. Oceanogr. Works Jap., Vol.11, 65-72

Yeo, R.R. (1964) Life History of Common Cattail. <u>Weeds</u>, Vol.12, 284-288

Reddy, K.R. and DeBusk, T.A. (1987) State-of-the-Art Utilization of Aquatic Plants in Water Pollution Control. <u>Wat. Sci. Technol.</u>, Vol.19 (10), 61-79

APPENDIX A.

The following co-authored papers were published during the course of the research project:

A2 - A8 The Use of Macrophytes for Water Pollution Control in China and the UK. IAWPRC Specialist Group: The Use of Macrophytes in Water Pollution Control, <u>Newsletter</u> No.3, March 1990, pp 25 - 31

A8 - A17 Metal Uptake and Associated Pollution Control by <u>Typha latifolia</u> in Urban Wetlands. In: P.F. Cooper and B.C, Pindlater (Edits) Proceedings of International Conference on <u>Use of Constructed Wetlands in Water</u> <u>Pollution Control</u>, September 1990, pp 451 - 459